OPEN ACCESS

Sensors ISSN 1424-8220 www.mdpi.com/journal/sensors

Review

Fully Integrated Biochip Platforms for Advanced Healthcare

Sandro Carrara ^{1,*}, Sara Ghoreishizadeh ¹, Jacopo Olivo ¹, Irene Taurino ¹, Camilla Baj-Rossi ¹, Andrea Cavallini ¹, Maaike Op de Beeck ², Catherine Dehollain ¹, Wayne Burleson ³, Francis Gabriel Moussy ⁴, Anthony Guiseppi-Elie ^{5,6} and Giovanni De Micheli ¹

- ¹ École Polytechnique Fédérale de Lausanne (EPFL), CH-1015 Lausanne, Switzerland; E-Mails: seyedehsara.ghoreishizadeh@epfl.ch (S.S.G.); jacopo.olivo@epfl.ch (J.O.); irene.taurino@epfl.ch (I.T.); camilla.baj-rossi@epfl.ch (C.B.-R.); andrea.cavallini@epfl.ch (A.C.); catherine.dehollain@epfl.ch (C.D.); giovanni.demicheli@epfl.ch (G.D.M.)
- ² Interuniversity Microelectronics Centre (IMEC), B-3001 Leuven, Belgium;
 E-Mail: opdebeem@imec.be
- ³ Department of Electrical and Computer Engineering, University of Massachusetts, Amherst, MA 01003, USA; E-Mail: burleson@ecs.umass.edu
- ⁴ Brunel Institute for Bioengineering, University of Brunel, West London, UB8 3PH, UK; E-Mail: moussyf@who.int
- ⁵ Department of Electrical and Computer Engineering, Center for Bioelectronics, Biosensors and Biochips, Clemson University, Anderson, SC 29625, USA; E-Mail: aguisep@clemson.edu
- ⁶ ABTECH Scientific, Inc., Richmond, VA 23219, USA
- * Author to whom correspondence should be addressed; E-Mail: sandro.carrara@epfl.ch; Tel.: +41-21-693-0915; Fax: +41-21-693-0909.

Received: 19 June 2012; in revised form: 10 July 2012 / Accepted: 17 July 2012 / Published: 8 August 2012

Abstract: Recent advances in microelectronics and biosensors are enabling developments of innovative biochips for advanced healthcare by providing fully integrated platforms for continuous monitoring of a large set of human disease biomarkers. Continuous monitoring of several human metabolites can be addressed by using fully integrated and minimally invasive devices located in the sub-cutis, typically in the peritoneal region. This extends the techniques of continuous monitoring of glucose currently being pursued with diabetic patients. However, several issues have to be considered in order to succeed in developing fully integrated and minimally invasive implantable devices. These innovative devices require a high-degree of integration, minimal invasive surgery, long-term biocompatibility, security and privacy in data transmission, high reliability, high reproducibility, high

specificity, low detection limit and high sensitivity. Recent advances in the field have already proposed possible solutions for several of these issues. The aim of the present paper is to present a broad spectrum of recent results and to propose future directions of development in order to obtain fully implantable systems for the continuous monitoring of the human metabolism in advanced healthcare applications.

Keywords: biochip; CMOS design; enzymes, biotechnology; nanotechnology; potentiostats; biocompatible membranes; security; privacy; remote powering

1. Introduction

According to the Molecular Diagnostics Survey Reports [1], diagnostics testing influences approximately 70% of health care decisions. This means that diagnostics are essential tools for diagnosing and managing numerous health care conditions, ranging from infectious diseases to non-communicable diseases such as diabetes. In fact, non-communicable diseases, or NCDs, are by far the leading cause of death in the world, representing 63% (36 million) of all annual deaths [2].

New diagnostic and monitoring technologies are needed across the economic spectrum, however, since some 80% of all NCD deaths occur in low- and middle-income countries, it is important to address some difficult challenges such as the need for low cost, low power consumption and robustness. Many of these devices should also be usable in places where there are few trained health care workers, and where parts, supplies and maintenance are very difficult to obtain. Fortunately, many of the diagnostics developed for the poorest countries could also have a positive impact in high income countries because their lower cost can help address the rapidly increasing health care costs both for individuals and for national health care systems. Fully integrated biochip platforms could significantly reduce the need for expensive pieces of equipment and laboratory procedures.

Fully integrated biochip platforms can build on existing technologies. Continuous monitoring is already in the market with commercially available devices for glucose [3] and lactate [4] while experimental prototypes have been already proposed for other endogenous metabolites, like glutamate [5] and ATP [6], as well as for exogenous metabolites (typically therapeutic compounds), like cyclophosphamide and naproxen [7]. The technology for continuous monitoring of glucose is robust enough to provide sensors with life-time up to 8 months when implanted in mice [8] and one year in pigs [9]. Sensor performance has been improved to meet sensitivity on physiological concentrations in human blood by using carbon nanotubes in case of both endogenous [10] and exogenous [7] metabolites.

The next step in this development is to hierarchically integrate all of these metabolite sensors in compact and robust biochip platforms to enable minimally invasive human telemetry with negligible costs. Two significant challenges are being addressed: extremely small chip sizes are required to be minimally invasive; and implementation in conventional CMOS manufacturing process is required to assure extremely low per unit fabrication costs. Several additional challenges arise in biosensors, including auxiliary sensors, analog circuit design, remote powering schemes, and security and privacy concerns. Temperature and pH [11] sensors are necessary in the platform for calibration goals. Fully

on-board generation of voltage ramps [12] is required to operate the sensing mechanism in the case of drugs [7]. Remote powering is required to address battery-less implants in reducing device size and cost [13]. The personal nature of the data created by biosensors, along with the fact that human health and safety are at stake, requires that security and privacy issues being considered early in the design. Although biosensors can be viewed as just another type of sensor and thus can build on the large literature related to sensor design and sensor networks, the fact that the data is coming from a human being changes the assumptions and threat models. Moreover, the *in vivo* environment presents unique challenges related to calibration, the foreign body response and signalling. These issues will be critical to the acceptance of these technologies by patients, physicians and regulatory bodies, as well as to their responsible deployment in the field.

This paper presents a bottom-up review of some recent advances in these domains and to propose and discuss a comprehensive approach to develop fully integrated biochip platforms for advanced healthcare. Section 2 presents recent advances in using nano-materials (carbon nanotubes, gold nanoparticles, silicon nanowires) to enhance sensor performance in terms of detection limit and sensitivity; Section 3 proposes several proteins (oxidases and cytochromes) used in order to provide specificity to a biochip; Section 4 describes the integration of nano and bio materials onto silicon platforms in order to develop multi-panel arrays; Sections 5 and 6 address the issues related to microelectronics for the biosensing front-end as well as for the problem of remote powering of implanted battery-less devices; Section 7 describes the issues of biocompatibility that are of special importance in developing fully-implantable devices; Section 8 addresses the unique concerns of security and privacy that arise in biosensors that transmit data wirelessly; Section 9 envisages future developments and applications of biochip based platforms connected with geographical networks for fully integrated and advanced healthcare; Section 10 concludes the paper.

2. Nano-Materials to Enhance the Sensors Performance

2.1. Materials Selection for Advanced Biosensors

Biosensors are based on the principle of converting a biochemical quantity into an electrical signal through the use of electrodes. Currently, a wide variety of different materials are used for the preparation of electrode surfaces for biosensing applications [14]. Most commonly used are glass and other oxide surfaces because of their favourable characteristics [15]. Also widely used are gold [16], micro-porous gold, graphite [17], glassy carbon [18] and indium tin oxide (ITO) [19].

2.1.1. Surface Materials and Modifications

An increasing number of sensing applications use screen-printed electrodes. Screen-printed electrodes (SPEs) are devices that are produced by printing different conductive inks on various types of insulating plastic or ceramic substrates [20]. Graphite materials are preferred due to their simple technological processing and low-cost [21]. Metals such as gold and silver are also used in the construction of SPEs for the analysis and determination of various analytes [22]. In most cases, the working electrode consists of a thin film made by Hg, Au, Ag, Ni or Bi, applied to the graphite surface of the electrode. Renedo *et al.* [8] has reviewed recent developments of screen-printed electrodes and

their applications. Another class of materials used for the fabrication of electrochemical biosensors are conductive polymers [23]. Guiseppi-Elie *et al.* has reviewed the use of conductive electroactive polymers in biosensors [24].

2.1.2. Surface Nanostructures

The selection and development of an active material for the electrode is a challenge. The active sensing materials should enable the biological recognition elements (biomolecules, usually proteins or enzymes), to act as a catalyst for sensing a particular analyte or a set of analytes. Recent developments in nanotechnology have revealed several new nano-materials, which have useful properties for numerous electrochemical sensor and biosensor applications [25,26]. By using nanostructures, it is possible to control the fundamental properties of electrode materials and enhance the electron transfer between the electrode and the enzyme, thus improving the catalytic reaction [27,28]. The enhancement of electron transfer is an extremely important challenge in the case of enzymatic biosensors, because a protein shell electrically insulates the redox-active site of most enzymes. Nano-materials such as carbon-nanotubes or nanoparticles have a promotion effect on the direct electron transfer between enzymes and electrode surfaces, thus obviating the need for mediators or co-substrates [29–31].

Many nano-materials (nanotubes, nanowires, nanoparticles, polymers, grapheme, quantum dots), have been used as intermediate layers for integrating electrodes with biomolecules (enzymes, proteins, antibodies, *etc.*), aiming the development of electrochemical sensors for the detection of metabolites and drug compounds [27,28].

2.2. Carbon-Nanotubes

Carbon nanotubes (CNTs) have been recognized as very promising nanomaterials for enhancing electron transfer [32] in biosensing (Figure 1) thanks to their electrical and electrochemical properties which make them suitable to be integrated into biological sensors. For these applications, carbon nanotubes present several advantages: small size with larger surface area, high conductivity, high chemical stability and sensitivity [33], high electrocatalytic effect and a fast electron-transfer rate [34].

Recent studies have demonstrated that CNTs enhance the electrochemical reactivity of proteins or enzymes with retention of their biocatalytic activity [32,35]. The nanotubes and enzyme molecules are of similar dimensions, which facilitate the adsorption of the enzyme without significant loss of its biocatalytic shape, form or function. Carbon nanotubes (CNT) have been extensively studied because of their unique structure-dependent electronic and mechanical properties, which make them suitable for many different electrochemical sensing devices, ranging from amperometric enzyme electrodes [28] to DNA hybridization biosensors [27]. The role of CNTs for the construction of novel biosensing devices has been recently reviewed [27,32,36,37].

In order to integrate biomolecules with CNTs, chemical/electrochemical treatments have to be realized for the introduction of oxygenated functionalities such as hydroxyl groups, which provide sites for covalent linking of biomolecules [27].



There have been a number of approaches to physically adsorb CNTs on electrodes by dispersing in a binder such as Nafion [38], by forming the nanotube equivalent of a carbon paste which can be screen printed, by forming composites with conductive polymers, or by drop casting a solution of CNTs (in solvents such as dimethylformamide [39]) onto an electrode without any binders. Also chitosan has been used for the dispersion and adsorption of CNTs [40–42]. With physical adsorption, the resultant electrode had randomly distributed tubes with no control over the alignment or orientation of the nanotubes. Self-assembly techniques, where aligned nanotubes are directly grown off a surface, were used in order to control the alignment of CNTs [42].

2.3. CNT-Hybrid Materials

Hybrid composite materials, based on integration of CNTs with other nano-materials, exhibit special properties due to the synergic effect from the individual components [27]. In particular, due to large improvement of the electron transfer in case of both nanoparticles [43] and nanotubes [44].

Carbon nanotubes–nanoparticle composites have demonstrated an enhancement in the electrocatalytic efficiency of many electrochemical processes [45–47]. Gold nanoparticles [48], metal alloy nanoparticles/multi-walled carbon nanotubes (MWCNTs) [49,50], encapsulated platinum nanoparticles onto MWCNTs [51], and MWCNTs with SnO₂ nanoparticles [52] are some recent examples.

Carbon nanotube-conducting polymer composites are one of the most widespread approaches for the preparation of electrochemical sensors [53–55]. The combination of the well-known characteristics of conducting polymers (good stability, reproducibility, strong adherence and homogeneity in electrochemical deposition) [56], along with those of CNTs, leads to improve sensing performance, also in presence of biomolecules (e.g., enzyme, proteins). Composites of conducting polymers and CNTs have been synthesized by either chemical or electrochemical polymerization [57–59]. Some recent applications are the incorporation of CNTs in polypyrrole-modified electrodes [60], electro-chemical sensors based on CNT-polyaniline-modified [61], and solubilization of CNTs in poly(vinyl alcohol) (PVA) [62].

Other composites involving carbon nanotubes have been recently developed: a composite film of MWCNTs and cyclodextrin (MWCNTs-CD) as an electrochemical sensor for the determination of adenine and guanine [63], and CNTs and room temperature ionic liquids (RTILs) composites [64]. CNTs incorporated in sol-gel matrices were developed for several electrochemical sensors [65–67].

2.4. Nanoparticles

Nanoparticles (normally with dimensions in the range of 1–100 nm) have unique chemical and electronic properties due to their small size that can be used to construct improved electrochemical sensors and biosensors. Different kinds of nanoparticles have been used in different electrochemical sensing systems, such as enzyme sensors, immunosensors and DNA sensors [68]. Generally, metal nanoparticles have excellent conductivity and catalytic properties, which make them suitable for enhance the electron transfer between redox centers in enzymes and electrode surfaces.

The main functions of nanoparticles can be summarized as: (1) their ability to facilitate biomolecule immobilization (mainly oxide nanoparticles); (2) catalysis of electrochemical reactions; (3) enhancement of electron transfer through increased surface area (mainly metal-nanoparticles—shown in Figure 2); (4) labelling of biomolecules (quantum dots [69]); and (5) acting as reactant [68,70,71].



Figure 2. Nanoparticle-mediated sensing.

Due to their large specific surface area and high surface free energy, nanoparticles can strongly adsorb biomolecules [72]. The adsorbed biomolecules can retain their bioactivity because of the biocompatibility of nanoparticles [73]. Since most of the nanoparticles carry charges and they can electrostatically adsorb biomolecules with different charges. Besides the common electrostatic interaction, some nanoparticles can also immobilize biomolecules by covalent linkage or through the entrapment in polymers. Metal nanoparticles (Au, Ag, Pt), oxide nanoparticles (SiO₂, TiO₂, ZrO₂, MnO₂, Fe₃O₄), or semiconductor nanoparticles (CdS, PbS), have been investigated in recent research and their applications have been newly reviewed [54,55].

2.5. Nanowires

Electrochemical sensor devices based on nanowires have been widely reported in the literature [26,74]. Different materials have been investigated for the fabrication of nanowires, such as gold [75], platinum and palladium [76], lanthanide hydroxide nanowires [77], metal-oxide nanowires [78], and silicon nanowires [79].

The material properties can be more precisely controlled by manipulating the conditions during nanowire synthesis, using well-developed doping techniques, and by suitable functionalization treatments [74] (e.g., antibodies in Figure 3).



2.6. Graphene and Other Carbon-Based Nanomaterials

Another very interesting nano-material for sensors application is graphene. Graphene has shown great promise in many electronics applications because of its unique physiochemical properties: high surface area, excellent thermal and electric conductivity and high mechanical strength [80]. Graphene-based electrodes have shown better enhancement in electrocatalytic activity than carbon nanotubes when used as electrode nanostructures. Several electrochemical sensors based on graphene and graphene composites for bioanalysis and environmental analysis have been developed [81,82]. A new graphene/AuNPs/GOD/chitosan composite-modified electrode was constructed through a simple casting method for a glucose sensor [83]. A single-layer graphene oxide was adsorbed on the 3-aminopropyltriethoxysilane (APTES)-modified conductive electrodes for the fabrication of a glucose sensor based on glucose oxidase [84]. Other carbon-based nanomaterials used for biosensor applications [85] are: (1) carbon nanotube paste electrodes [86]; (2) carbon nanotube nanoelectrode (based on CNT nanoelectrode ensembles) [34]; (3) carbon nanofibers [87]; (4) exfoliated graphite nanoplatelets [88]; and (5) highly ordered mesoporous carbon [89].

2.7. Conductive Polymers/Nanocomposites

Nano-structured conducting polymers and polymer composites have recently shown their potential applications in biosensors [90]. Conductive polymer nanowires (CPNWs) are an attractive alternative to silicon nanowires and carbon nanotubes because of their tuneable conductivity, flexibility, chemical diversity, and ease of processing [91]. CP–nanoparticles [92] and CP-carbon nanotubes composite materials [60–62] have also been investigated, due to their hybrid properties.



3. Molecular Recognition for Biosensors

The immobilization of a biomaterial onto a conductive support is the basic feature of a biosensor device and also the most critical step. Biomaterials that can be thus assembled include, enzymes [93], receptors [94] antibodies or antigens [95], oligonucleotides or DNA fragments [96,97], or low molecular weight molecules exhibiting affinity interactions with other biomaterials such as cofactors, (e.g., NADPH [98]), or proteins as biotin [99]. Systems that employ enzymes as bioactive interfaces represent the most extensively studied assemblies in biosensors. The high specificity of enzyme-substrate interactions, and the usually high turnover rates of biocatalysts, opens the way to tailor sensitive and specific enzyme-based biosensor devices [100].

Enzymes are proteins that act as powerful catalysts to convert substrates into products [34–40]. Some enzymes, also known as redox-enzymes, catalyse reactions that produce or consume electrons [93]. Direct electrical activation of enzymes, and particularly redox enzymes, represents a general approach to stimulate the oxidation (or reduction) of the enzyme substrates. Electron transfer between the electrode and the redox enzyme should be fast, and the resulting current corresponds to the turnover rate of the electron exchange between the substrate and the enzyme.



Figure 4. Sensing principle of oxidases.

Hence, the transduced current (detected with amperometric techniques [14]), reflects the substrate concentration in the system [101]. The functionality of biosensors relies on the electrical contact between enzymes and electrode surfaces. Biomaterials and biomolecules must be assembled on solid conductive supports providing an appropriate electronic communication between the biological matrices and the supporting element [100].

In general, many oxidoreductases including glucose oxidase, catalyse the oxidation of substrates by electron transfer to oxygen to form hydrogen peroxide [14] (Figure 4). These oxidoreductase enzymes can be immobilized on conducting polymer surfaces and the H_2O_2 formed in the reaction is measured amperometrically [57]. The most commonly used enzymes in biosensing are glucose oxidase (GOx) [102] and horseradish peroxidase (HRP) [16]. Other, less commonly used enzymes comprise beta-lactamase [103], and urease [104].

Several studies on amperometric biosensors have been made for the development of drug monitoring-systems, such as the use of the enzyme aryl-acylamidase for assays of paracetamol in connection with the detection of the liberated aminophenol, the use of theophylline oxidase and a ferrocyanide mediator for the biosensing of theophylline, or the immobilization of salicylate hydroxylase for the detection of salicylate in the presence of NADH and dissolved oxygen. Organic-phase tyrosinase and peroxidase electrodes have been used for the monitoring of phenolic and peroxide antiseptics in anti-infective pharmaceutical formulations [105].



Figure 5. Sensing principle of P450 cytochromes.

Another potential approach is the use of cytochrome P450 enzymes (CYP), as the biological recognition element in amperometric biosensors for drugs monitoring [106–109] (Figure 5). Recently Sharma *et al.* [110], extensively reviewed enzymes used in development of biosensors, also for industrial applications.

Various methods are available for immobilization of enzymes, but not always appropriate for manufacturing biosensors. Several problems relating to the functioning of an enzyme system must be accounted for, like loss of enzyme activity and maintenance of enzyme stability. An efficient enzyme deposition method must satisfy the following requirements [111]:

- The immobilization of the enzyme on transducer surfaces must be stable and provide an efficient electron transfer;
- The enzyme must completely retain its biological activity (and enzyme stability);
- The enzyme must be accessible when immobilized;
- The material used for immobilizing the enzyme must be compatible and chemically inert towards the host environment.

Immobilization of biomolecules can be carried out using many different procedures, like adsorption, entrapment, intermolecular cross-linking and covalent binding [14,110,112,113].

3.1. Adsorption

The adsorption process can be classified as physisorption, chemisorption or electrostatic attraction. Binding forces are mainly due to hydrogen bonds, polar and hydrophobic interactions, Van der Waals forces and formation of electron transition complexes [93]. Which intermolecular forces exactly take part in the interaction will depend on the particular protein and surface involved. The resulting layer is likely to be heterogeneous and randomly oriented, since each molecule can form many contacts in different orientations for minimizing repulsive interactions with the substrate and previously adsorbed proteins [96]. However, biosensors with adsorbed enzymes suffer from poor operational and storage stability, so that adsorption in combination with cross-linking has also been used for the immobilization of enzymes [93]. Intermolecular cross-linking of biomolecules are made by reagents such as glutaraldehyde [41], other bi-functional or multi-functional reagents (hexamethylene di-isocyanate, 1,5-difluro-2,4-dinitrobenzene and bisdiazobenzidine-2,20-disulphonic acid, *etc.* [93]) and done in the presence of filler molecules such as bovine serum albumin (BSA) or polyethylene glycol.

Enzymes and nanomaterials are usually dispersed in solvents by sonication, resulting in a suspension. An aliquot is dropped on the electrode surface, followed by air drying, producing the modified electrode. Several solvents can be used for dispersion of nanomaterials and production of electrochemical biosensors, for example dimethylformamide and cyclohexanone [114], chloroform or charged polymers, such as chitosan and nafion. Chitosan is a polysaccharide with good biocompatibility, soluble in slightly acidic solution due to protonation of amino groups and presenting film-forming ability due to insolubility in solution with pH above pKa (6.3). Chitosan has been used in several sensing application as a dispersant agent [115–117]. Electrodeposition of nanostructures and enzyme in chitosan has been successfully performed for the fabrication of sensors as reported in [40,118–120]. Another interesting dispersant is nafion, a perfluorinated sulfonated cation exchanger polymer, showing chemical inertness, thermal stability, mechanical strength, and antifouling property [72]. Nafion is used for the dispersion of nanostructures, in particular, of carbon nanotubes [121–123].

3.2. Covalent Bonding and Bioaffinity Immobilization

Modification of electrode surfaces by covalent bonding is advantageous, since it produces a more stable material, resulting in an irreversible binding and a high surface coverage [100], but it requires a high amount of bioreagent and is generally poorly reproducible [93] as such bonding has the potential to adversely affect the active site. Covalent binding is accomplished through functional groups of exposed amino acids of the enzyme, which are not essential for its catalytic activity, such as amino, carboxylic, imidazole, thiol, and hydroxyl groups. These groups create covalent linkage with the functional groups that are available on the surface of the solid support, or that can be generated by chemical or electrochemical treatment. Many procedures have been reported to covalently link an enzyme to a support [100,112].

Immobilization of enzymes can also be achieved by affinity interactions between functional groups such as lectins, (strept)avidin, or metal chelates on an activated electrode surface and an affinity tag such as carbohydrate residue, biotin, histidine or cysteine which is present or genetically engineered by Site Directed Mutagenesis (SDM) at a specific location in the protein sequence [72].

Affinity interaction is stronger than physical adsorption, even if it is not as strong as the covalent linkage, and it provides a basis for a controlled and oriented immobilization of the enzyme on different supports [93,124].

3.3. Physical Entrapment

Polymers have gained a remarkable position in the biomedical field as materials for fabrication of various devices and for tissue engineering applications. Polymer films have good stability, reproducibility, more active sites, homogeneity in electrochemical deposition and strong adherence to electrode surfaces [57]. Electropolymerization has received great attention in recent years as a good approach to prepare immobilization matrices for biosensors [56], since it is possible to precisely control film thickness, permeation and charge transport characteristics by the adjustment of the electrochemical parameters. The modification of surfaces with polymeric films has been used in the development of biosensors to protect the surface of the electrodes from impurities, block interfering, incorporating mediators and provide biocompatibility [58].

3.3.1. Conductive Polymers

Conducting electroactive polymers (CEPs) [57,59,111] have attracted much interest as suitable matrices of biomolecules and can be used to enhance enzyme stability and to provide a suitable environment for their immobilization. The techniques of incorporating biomolecules into conducting polymeric films permit the localization of biologically active molecules on electrodes of any size or geometry and are particularly appropriate for the fabrication of multi-analyte amperometric biosensors.

Electrically conducting polymers are known to have considerable flexibility in chemical structures, which can be modified as required. Another advantage offered by conducting polymers is that the electrochemical synthesis allows direct deposition of a polymer on the electrode surface while simultaneously entrapping the protein molecules. Conductive polymers can be reversibly doped and undoped using electrochemical techniques accompanied by significant changes in conductivity. Conducting polymers have the ability to efficiently transfer electric charge produced by a biochemical reaction [57] to an electronic circuit [111].

Conducting polymers belonging to polyenes and polyaromatics such as polyacetylene, polyaniline (PANI), polypyrrole (PPy), polythiophene (PTh), poly (*p*-phenylene), poly(phenylene vinylene) classes have been extensively studied [125]. Recent applications in the field of analytical sensors based on conducting polymer are reviewed by Ahuja [111], Malhotra [57], and Guiseppi-Elie [59].

Biomolecules can be entrapped in conductive polymers through physical adsorption [111,126,127], cross-linking with glutaraldehyde [41], and covalent attachment [128,129], but the most reproducible and controlled technique is the electrochemical polymerization. Entrapment of the enzyme involves electrochemical oxidation of the monomer in the presence of the enzyme to form a polymer, which incorporates the homogenously distributed enzyme molecules during its polymerization reaction. This method is easy to perform and allow a simple one-step fabrication of the electrodes in which enzyme, mediators and additives can be simultaneously deposited simply by adding them to the polymerization solution [111,128].

3.3.2. Enzyme Entrapment/Encapsulation

Enzymes for biosensors have been entrapped in membranes based on advanced materials, including hydrogels (which contain cryohydrogel, organohydrogel and grafting copolymer), sol-gel-derived organic-inorganic composites, and lipid membranes [130].

Hydrogel, based on a cross-linked network of polymers able to swell with water, is one of the ideal enzyme-immobilization materials, providing a biocompatible microenvironment for the enzyme to maintain its natural configuration [131], and at the same time mediating electron transfer between the enzymes and the electrode via electron diffusion through mobile polymer segments [14]. The most used hydrogels for immobilization of enzymes are poly(vinyl alcohol) (PVA) [132], or PEG (polyethylene glycol), which prevents nonspecific adsorption of biomolecules [133]. A kind of polyhydroxycellulose (PHC), a mixture of PVA and carboxymethylhydroxyethyl cellulose, was investigated as a novel immobilization method for enzymes [130].

Sol-gel processes form sol-gel-derived organic-inorganic composites. They involve hydrolysis and co-condensation of silane precursors like tetramethoxysilane (TMOS) and tetraethylorthosilicate (TEOS), with acid or base catalysis. The material formed provides a promising means for immobilizing bioactive molecules [134]. Further, addition to the starting sol of polymers, e.g., poly-dimethyl-siloxane, polyamides, polyacrylates, polyethylene glycol (PEG), and poly(ethylene-oxide), was investigated to regulate the inorganic condensation–polymerization process [130]. Polyaniline (PANI) was electrochemically entrapped onto TEOS-derived films on indium tin oxide (ITO) coated glass to immobilize lactate dehydrogenase (LDH) for lactate sensing [135]. Acetylcholinesterase (AchEs) enzyme in TMOS+MTMOS+PEG-derived sol-gel has been immobilized on a modified screen printed electrode for the detection of carbamate insecticides [136].

Lipid membranes, imitating the basic structure of nature's biomembrane, represent a relatively biocompatible structure for the development of new types of electrochemical sensors [130]. Lipid membranes can be formed by casting solutions of lipids or surfactants in organic solvents or of aqueous vesicle dispersions onto solid surfaces. Recently tethered bilayer lipid membranes (tBLM) have been developed. In tBLMs, the bilayer is attached to the surface via special chemical anchors, which on one side are bound to the surface and on the other side insert into a bilayer leaflet. For this purpose, lipid derivatives (phospholipids, cholesterols, alkyl chains or phytanyl groups), have been synthesized, which, via a hydrophilic linker, are connected to a group that binds strongly to the surface [137,138].

Dialysis (ultrafiltration) membranes have also been used [139], to entrap the enzyme between the electrode and a dialysis membrane having a cut-off inferior to the size of the enzyme, while the substrate is able to freely diffuse through the dialysis membrane. Such an immobilization method allows small amounts of enzyme to be used and the integrity of the enzyme to be preserved [93]. Immobilization layers based on clay-materials were also investigated in some sensing applications [140].

3.4. Langmuir-Blodgett, Layer-by-Layer, Self-Assembled-Monolayer (SAM)

Two of the most used methods for the immobilization of biomolecules on electrodes are the electrostatic layer-by-layer (LbL) and the Langmuir-Blodgett techniques, which are complementary to each other in terms of the types of material that can be employed.

3.4.1. Langmuir-Blodgett Based Electrochemical Biosensors

The Langmuir-Blodgett (LB) technique promotes a high control of the physical and chemical properties of nanostructured organic films and it has been widely used for the production of miniaturized devices for enzyme immobilization. The LB technique is based on the transfer of insoluble monolayers supported on liquid-air interfaces to solid supports that intercept vertically the liquid surface. The material can form a monomolecular film at the liquid-air interface in such a way that hydrophobic moieties are oriented towards the air and the hydrophilic ones towards the aqueous phase. This film, usually called Langmuir film, can be compressed through liquid barriers in order to obtain the desired surface density [72].

The LB technique is a suitable method for immobilization of enzymes because of its well-ordered structure of the ultrathin films and its ability to control the amount of biomolecules immobilized by changing the number of deposited layers. The classical materials used to form Langmuir and LB films are lipids, such as fatty acids, glycerophospholipids, ionic liquids [141], synthetic polymers [142], proteins [143], nucleic acids [144], and polysaccharides [145]. Girard-Egrot recently reviewed applications of enzyme with lipidic Langmuir–Blodgett films [146].

Electrochemical sensors based on the LB technique contain enzymes such as horseradish peroxidase [147], hemoblogin [148], glucose oxidase [149], acetylcholinesferase [150], urease [151], tyrosinase [152], choline oxidase [153], lactase and galactose oxidase [154], uricase [155], for detection of several substances including hydrogen peroxide, glucose, choline, urea, phenols, lactose and uric acid.

3.4.2. Layer-by-Layer (LbL) Based Electrochemical Biosensor

The LbL technique has been found to be a useful way for assembling a number of organic and inorganic substances, including proteins, DNA, viruses, dendrimers, and nanoparticles [156]. Basically, the processes of film fabrication by the LBL technique are governed by the adsorption of organic polyelectrolytes (PE) with opposite charges present on their molecular structure, in such a way that film thickness, porosity and morphology can be precisely controlled according to the application to a target. An important advantage in the use of the LbL technique to construct biosensors is the possibility to incorporate inorganic materials and biomolecules [157].

Multilayer films are grown by alternate dipping of the modified electrodes into the enzyme solution (or nanoparticles solution) and a PE solution, respectively. Polycations that have been used in LbL films formation include poly(allylamine) (PAA), poly(L-lysine) (PLL), poly(ethyleneimine) (PEI), poly(dimethyldiallylammonium chloride) (PDDA), poly(allylamine hydrocholoride) (PAH) and chitosan (CHIT). The most commonly used polyanions are poly(stryrenesulfonate) (PSS), poly(vinylsulfonate) (PVS), poly(anilinepropanesulfonic acid) (PAPSA), poly(acrylic acid) (PAA) and poly (methacylic acid) (PMA) [157,158]. Water-soluble proteins such as cytochrome c (Cyt) [159], horseradish peroxidase (HRP) [160], myoglobin (Mb), lysozyme (Lys), histone f3, hemoglobin (Hb), glucoamylase (GA), and glucose oxidase (GOx) have been incorporated in LbL and used for biosensors [158,161].

However, in the development of electrochemical biosensors based on LBL assembly methods, other strategies such as biological affinity, covalent binding, cross-linking, and electrodeposition have also been used, and also multi-strategies have been investigated [158].

3.4.3. Self-Assembled-Monolayer (SAM)

Redox proteins can be adsorbed via electrostatic interaction or covalently immobilized on Self-Assembled-Monolayers (SAM) [162], which can lead to a better-controlled electron transfer between the protein and the electrode [163]. With SAMs it is possible to regulate the distance between the enzyme binding-site and the electrode surface, changing the chain length (the so called 'tail'), of SAM, which allows the film to be tightly packed and oriented on the surface thanks to the weak interactions between the tails. Usually alkane-thiol or other thiol-terminated chains covalently bind the self-assembled molecule to the metal surface of the electrode. At the other end of the chains, a group is able to specifically interact with a group on the protein surface, thus adding selectivity to the modified electrode for particular proteins.

4. Integration of Bio and Nano Materials on Multi-Electrode Platforms

In the last few years the use of miniaturized electrodes fabricated by thin-film technology has become more and more common in sensor research because it gives some advantages (e.g., smaller background current and faster time response). The combination of microelectrode devices with the immobilization of biochemical compounds such as enzymes provides excellent prerequisites for the development of miniaturized biosensors in order to monitor many metabolites in parallel [164]. These biodevices should be composed of working electrodes that share the same counter and reference electrodes. Working electrodes have to be functionalized first with MWCNTs to increase sensitivity and decrease the detection limit (Figure 6).



Figure 6. Multi-array platforms and CNT integration.

Then, the enzymes have to be incorporated to enhance the selectivity for each metabolite (Figure 7). A fixed or variable potential between a reference electrode and working electrodes must be applied according to the functionalization of each electrode.





4.1. MWCNTs Integration Methods

4.1.1. Direct Growth onto Chip

Direct growth by Chemical Vapour Deposition (CVD), is one of the most used forms of growing CNTs. In CVD, a hydrocarbon gas with carbon atoms is provided for the CNT growth on metallic catalysts [165]. The catalyst can be introduced in a vapour phase or is present as a solid layer on the substrate on which CNTs have to grow (Figure 8). Typically, nanotubes are grown on semiconducting or usually insulating silicon, silicon dioxide, quartz. The whole setup is then placed in a quartz tube, maintained at atmospheric pressure in a flow furnace. The hydrocarbon gas mixture is passed over the quartz at high temperatures (500–1,000 °C). The gas catalytically decomposes over the metal particles (iron or nickel) at these temperatures. Compared with other methods of CNT production, CVD has been regarded as the most promising for industrial application. In particular, CVD is the preferred choice to grow patterned CNTs on metal catalyst particles or islands deposited on top of the substrates. These catalytic metals are seeded onto metallic tracks realized on a substrate through standard microfabrication techniques (standard UV photolithography, electron-beam evaporation, lift-off).

Figure 8. CVD growth of CNTs.



For many applications, such as biosensing, high electrical conductivity is required usually needing a conductive substrate to connect with CNTs. It is challenging to grow CNTs on metal substrates since the elevated temperature may activate the diffusion of the catalyst material into the substrate, which inhibits its activity. Catalysts tend to form alloys with substrates or to coarsen and segregate during growth [166,167].

To overcome this problem, one possible strategy is to employ pure catalytic metal [168] or alloys [169] containing at least one of these well known catalyst materials as substrates. However, if the substrate selection is constrained to specific metals an alternative effective choice is to deposit a thin insulating layer to stop the poisoning of catalyst particles. Insulating layers, however, cannot provide an electrical connection to the conducting substrate, and additional fabrication processes are required to electrically contact with the substrate [166]. Nevertheless, some researchers succeeded in growing CNTs directly on metal substrates by using vapour pressure catalyst delivery [170]. This technique cannot be applied to the patterned growth approaches needed for biosensing applications.

It is a further challenge to grow CNTs on metallic substrates at CMOS compatible temperatures (400–450 °C) [171]. Lowering the growth temperature has frequently proven to be ineffective because the structural quality of CNTs usually degrades as reaction temperatures are decreased. Moreover, higher temperatures are often required to form the metal nanoparticles necessary for CNT growth.

Initial approaches to overcome these drawbacks consist of improving the CVD recipes by varying the reactant gases, pre-treatment and immobilization method of the catalyst and other deposition parameters. Up to now only few papers have reported this kind of approach for the realization of enzyme biosensors. Lin *et al.* [172] reported a glucose biosensor based on CNT nano-electrode ensembles made of low-site density aligned CNTs grown on a chromium-coated silicon substrate by plasma-enhanced CVD and using nickel nanoparticles as a catalyst. An epoxy-based polymer was then spin-coated and the protruding parts of the CNTs were removed by polishing. Then, glucose oxidase was immobilized directly on the broken tips of CNTs for glucose detection. In addition, Tominaga *et al.* [173] studied the direct electron transfer reactions of D-fructose dehydrogenase immobilized onto MWCNTs for D-fructose detection. The CNTs were synthesized by CVD method using iron nanoparticles derived from ferritin molecules and adsorbed onto the platinum plate.

CVD incorporation of CNTs not only improves the electrical contact between the active sensing material and the conducting substrate, but also ensures that the sensor is free of any other materials used as surfactants and/or binders [174]. However, due to the CNT hydrophobicity, CVD-CNTs need to be chemically and/or electrochemically treated for the easy incorporation of the enzymes. Finally, CNTs grown by CVD may collapse in the presence of solvents.

4.1.2. Covalent Bonding

The advantage of covalent bonding is linked to the great chemical affinity between gold and thiol and the good stability of the Au–S bond. Common to these methods is the formation of carboxylic acid groups on the surface of the CNTs, usually introduced by strong acid oxidation. They occur predominantly at the more reactive (open) ends or at defect sites of CNTs. Exploiting the high chemical reactivity of this functional group, it is possible to covalently attach biomolecules obtaining in this case a stable grafted surface that could be particularly attractive for biosensing purposes [175].

Typical approaches reported in the literature to link covalently CNTs onto gold include Au-S bonding and surface condensation. The first strategy mimics the formation of alkanethiol SAMs on a gold surface. Carboxyl-terminated MWCNTs must be modified with 2-aminoethanethiol and vertically self-assembled on gold electrodes. Substrate dipping in a suspension of functionalized CNTs such as ethanol [173], chitosan and tween [176] allows them to be vertically assembled onto the gold surface. With the latter method, the carboxylic functionalized MWCNTs are usually dissolved in an organic solvent and the gold substrates must be pre-modified with amino groups [177]. The surface condensation reaction is carried out with 1-ethyl-3-(3-dimethylaminopropyl)carbodiimide hydrochloride (EDC) as coupling agent.

In a mixed procedure, Kim *et al.* [178] added alkanedithiols to the surfaces of MWCNTs and immobilized them on SAM-modified Au substrates by dipping these substrates into DMSO solution containing MWCNTs. The adsorption of MWCNT-SH onto SAMs of alkanethiols takes place selectively depending on the chain lengths of the thiolate in the SAM.

One limitation of such integration is that the highly conductive CNTs were not in direct contact with the electrode surface and thus electron transfer could be hindered. In addition, different methods to link covalently SWCNTs-Au have been well established [179] especially for biodetection. However, to the best of our knowledge, these studies have not been published for MWCNTs despite their better electron conduction properties. It is due to the following challenges:

- MWCNTs are more difficult to disaggregate and suspend in solvents than SWCNTs.
- The common procedure employed to introduce thiol moieties selectively on the tips of the SWCNTs also modifies the sidewalls of MWCNTs inducing in most cases their random orientation onto the gold electrodes.

4.1.3. Micro-Spotting

Micro-spotting casting immobilization is the most widely used method for fabricating CNT-based electrochemical sensors (Figure 9). With this approach, MWCNTs must be dispersed in a certain solvent with sonication after their purification and activation pre-treatments. The resultant suspension must be cast onto the electrodes and allowed to dry. In addition to different solvents such as chloroform [10], various additives are added into solvents to assist the CNT dispersion (e.g., Nafion, polymers) [174]. The predominance of these methods is their simplicity, which makes them suitable to fabricate CNT-based electrochemical sensors.

Figure 9. Microspotting.



A microspotting system was developed for the rapid assembly of MWCNTs onto small surfaces [180–182]. In general, a micro-robotic station included three programmable components [180]:

- (1) a computer controllable X-Y-Z micromanipulator of the probe tips to appropriate location of them;
- (2) a computer controllable hydraulic pump system to inject CNT dilution through the capillary probes; and
- (3) a camera allowing an operator to locate microelectrodes.

The CNT suspension is transferred to the probe tip by the micro-pump. The probe is then commanded and moved to the desired position by using the computer controllable micromanipulator automatically. Finally, the CNT suspension is deposited onto the desired location [180].

CNT dilution spotting is challenging because of the difficulties in casting a very small CNT-based solution to particular positions automatically and precisely [181,179]. This technique was employed by Lai *et al.* to connect CNTs across two microelectrodes. CNTs were suspended in acetone [182] or ethanol [179,183] solutions. It was the first research group that employed and optimized the micro-spotting deposition process of CNTs by comparing the spot sizes, by using different probes and by studying the micromanipulator options to change the probe positions [179,183]. They tested also two different spotting methods to study the performance of the micro-robotic spotting system: injection method, spotting using a syringe pump to induce enough hydraulic pressure and droplet contact method, spotting by allowing a droplet to come in contact with substrate [180]. In the biosensing field, Boero *et al.* used a MWCNT solution based on Nafion. It was preferred to dispersion in chloroform since the last one is too volatile for the small volumes involved (400 pL). They performed automatic spotting of MWCNTs with a commercially available non-contact spotter. Multiple layers were spotted allowing Nafion solutions to dry in between two subsequent depositions [181].

5. CMOS Design for Highly Integrated Potentiostats

5.1. Circuit Design for Implantable Biosensor

To realize fully implantable and minimally invasive biosensor platforms, the sensor and the front-end electronics should be miniaturized and integrated into a single chip. Thus low power and low noise analog/mixed-signal integrated circuits must be carefully designed for all different types of measurement procedures. The required circuit elements to implement the front-end electronics can be divided into two main parts:

- (1) readout circuit;
- (2) voltage generator and potentiostat (Figure 10).

These parts will be introduced in the following subsections. The biosensor has 3 electrodes: working electrode (WE) reference electrode (RE) and counter electrode (CE).



Figure 10. Simplified frontend electronics.

5.1.1. Readout Circuit

The readout circuit serves to detect and amplify the sensor signal and convert it to a form which is can be modulated or transmitted. The readout circuit may include current mirrors, current amplifier, operational transimpedance amplifier (OTA), analog to digital converter (ADC), current peak detector, *etc.* As the miniaturized sensor has a very weak signal (pico- or pA), the readout circuit must be designed to have low noise and low leakage. On the other hand, as the whole system will be remotely powered, it has to consume low power, on the order of a few microwatts.

Many works have been done to design biosensor readout circuits. In the work of Ahmadi and Jullien [184] and Haider *et al.* [185], the sensor current was converted to frequency, which can be sent out by backscattering without any need for an ADC. However this complicates data processing steps such as error correction, storage or data encryption. In Li *et al.* [186] a switch-capacitor was used together with a synthesis method to suppress the noise and the offset of the amplifier. It amplifies a wide range of sensor current 100 fA–100 pA and converts it to voltage. Then, using an ADC, the voltage was digitized and can be modulated and sent out of the implanted device (Figure 11).

Figure 11. Schematic of the surface electrode array with the potentiostat and the readout circuit.



Another method that is introduced in [187] is to amplify the sensor current and digitize it using a current mode ADC [187] and [188] without converting it into voltage. This method doesn't need an OTA. The other advantage of the method is that the digitized signal can be processed and encrypted before being sent out. In yet another approach [189] used a low pass filter and an ADC to digitize the potential decay on a capacitor produced by the cell current.

5.1.2. Potentiostat and Voltage Generator

A potentiostat circuit has the purpose of keeping the reference and working electrode at a given potential to provide a stable bias voltage for measurement purposes. The stability of the potentiostat circuit should be studied carefully as the electrical model (*i.e.*, the RC equivalent model) of the biosensor can change significantly with the surrounding fluid concentration, temperature and also the applied voltage to the sensor. Therefore, to prevent any instability, a safety range in the phase margin of the amplifiers should be applied when designing the potentiostat, so that it does not become saturated by changes in the properties of the surrounding fluid (concentration, density, temperature, *etc.*) within the pharmacological range.

Another concern in designing the potentiostat is that the CE (Counter Electrode) voltage depends on the type of the sensor, cell current and the fluid properties. Thus, if the potentiostat limits the range of the CE voltage, it can potentially affect the accuracy of the measurements. Therefore, the potentiostat should be designed in such a way to let the CE change freely within the voltage range. This range can be acquired by modelling of the biosensor, or with some experiments prior to connecting the biosensor to the electronics.

The potentiostat can be designed using one of many amplifiers in different configurations [190,191]. Depending on the type of the measurement, a fixed or a variable voltage is needed to apply to the cell. For single-target chronoamperometry, the voltage is fixed and chosen on the basis of the electrochemical reaction. For cyclic voltammetry, this circuit sweeps repeatedly within the voltage range of interest. A band-gap reference circuit can be used to generate a fixed voltage [192]. To design a very slow triangular waveform voltage, a mixed signal design approach can be used (Figure 12). In [186], a digitally programmable waveform generator was used by employing an up-down counter, comparator, latch and a digital to analog converter (DAC).

Figure 12. Direct digital synthesizer (DDS) to generate a very slow ramp to drive a cyclic voltammetry measurement.



5.2. Multi-Target Sensing

The integration of multiple biotransducers into sensors [193] with the required electronics can pave the way to use smart systems in clinical diagnoses [194,195]. Therefore multiplexing circuits are required to support the readout of multiple current sources and the drive of multiple control points for the potential. Multiplexing can also be used to send out other data from auxiliary sensors such as the temperature and PH of the fluid as introduced in [11].

5.3. Different Noise Sources

The main sources of noise that can affect the system performance can be classified into three categories: (1) potentiostat and voltage generator noise; (2) current readout and amplification circuit; and (3) biochemical environment. The noise performance for the potentiostat and voltage generator block is not critical because the stimulus signal for amperometric interrogation is normally several hundred mill-volts, well above the noise level. However, in the current readout block, noise performance will significantly affect readout sensitivity because the response current of the biosensor could be very small (in sub-picoamp range).

The electrochemical noise includes the noise generated in the interface of the electrode, carbon nanotube, protein (enzyme) and the substrate. The low frequency component of the biochemical environment noise limits the sensitivity and determines the detection limit of the biosensor. So the noise from the biochemical environment should be minimized to enhance the biosensor performance. This is not straightforward, since the sensor noise is hard to quantify analytically, but it can be measured experimentally [164]. Figure 13 reports the Power Spectrum Density (PSD) registered with a sensor.



Figure 13. Noise PSD measured with screen-printed electrode at $V_{cell} = 650 \text{ mV}$.

It is an example of the measured low frequency noise of the biosensor during the detection of H_2O_2 . Some research has been done to model the noise of the carbon nanotube devices [196], neural sensors [197], and electrode-electrolyte interface [198]. However a closed form model for the biosensor noise has not yet been developed.

6. Remote-Powering

In realizing implantable biochip platforms, providing energy to the implant becomes a relevant issue if needed to decrease the impact on the patient in terms of pain and device size. Energy harvesting techniques might exploit natural or artificial power sources surrounding the person to provide energy to implanted circuits. The energy can be used directly or stored in capacitors or batteries [13,199]. Thus, batteries can either be eliminated or be smaller and with a longer lifetime. However, batteries may be avoided in order to decrease the biochip size with benefit on the patient's side. Remote powering is one of the best approaches if we would like to avoid batteries. In remote powering, an alternating current flows into an inductor in an external base station; and the resulting variable magnetic field generated induces an alternating current into one or more implanted inductors. In such cases, the system works in the near field region [200,201] where the distance between the external base station and the implant is smaller than the wavelength divided by two times pi ($\lambda/2\pi$) [202]. It exists in typical frequency bands for such applications, which are centered at 125 kHz, 6.7 MHz or 13.5 MHz [202].

The techniques of near field communications (NFC) and powering have been studied for several decades, and has been rapidly appearing in the marketplace for RFID tagging, mobile payments and other applications [203]. However, the miniaturization process of the implanted inductors to the millimetre level, while preserving the power efficiency, is still an open research topic. Moreover, it is not an easy task to maximize the total power efficiency of the remote powering link from the external base station to the output of the AC to DC converter, also called rectifier, as shown in [201]. This is due to the fact that this total power efficiency depends on several different blocks such as the Power Amplifier (PA) of the base station (in most cases a class E PA), the design of the primary coil of the base station, and of the secondary coil of the implant, the topology of the AC to DC converter, the voltage regulator, and others.

Inductive links present considerable additional complexity, when compared with other kinds of power transmission. Exploiting this technique, data can be transmitted from outside to inside the body (downlink) and *vice versa* (uplink) (Figure 14) without using a radio-frequency active transmitter which would otherwise consume significant power due to the RF oscillator and amplifier. This can be feasible by instead modulating the load of the secondary coil, varying in this way the total load seen by the primary coil. This technique of data transmission, often named load-modulation, allows significant energy savings by avoiding the use of an implanted RF active transmitter. Indeed, the RF active transmitter would otherwise have the highest power consumption among the components of an implantable biosensor. The capability to avoid an implanted RF active transmitter, together with a delivered power of up to a few milliwatts, makes this technique particularly suitable for very small (*i.e.*, minimally-invasive) implantable biosensors.

A wearable device to transmit power to an implantable sensor has been recently presented [204]. The prototype system, named IronIC Patch, is shown in Figure 15. The system is realized on a flexible FR4 substrate to be embedded into an adhesive skin patch and located directly over the implantation area. This placement reduces the probability of misalignments between the transmitting and receiving inductors, thus significantly improving the efficiency of the link for both power and data transmission.

Figure 14. Schematic representation of an inductive powering performing bidirectional data transmission.



Figure 15. Photo of the prototype IronIC Patch. The system can be placed over the implantation area with an adhesive bandage to power and communicate with implanted sensors.



The IronIC Patch can transmit power wirelessly to the implanted sensors by driving the transmitting coil with a class-E power amplifier. Downlink communication is achieved by means of an Amplitude Shift Keying (ASK) modulator. Uplink communication is achieved by monitoring the current drawn by the amplifier. The system also enables long-range communication with more powerful remote devices (e.g., mobile phone) by means of an embedded bluetooth module. Finally, the device is powered by two rechargeable lithium-ion polymer batteries. The system has been demonstrated to be particularly suitable to power implantable biosensors dedicated to glucose and lactate monitoring [184,205].

7. Biocompatibility

Biocompatibility is a critical issue that must be solved in order to implant electronics and sensors into the human body. Biocompatibility aspects cannot be neglected otherwise the implantation can cause important adverse effects to the body such as severe inflammation and necrosis of the tissue surrounding the implant in the short term, as well as long term systemic effects such as allergic reactions or even the onset of cancer. Furthermore, if a pronounced mismatch exists between the local body tissue and the implant, a pronounced 'Foreign Body Response' (FBR) will occur, causing walling of the implant by fibrosis after 3–4 weeks of implantation [206]. Due to the presence of this fibrotic tissue, chances are high that the sensor is not in contact with representative tissue/body fluids, causing an incorrect readout of the sensor. The preferential adhesion of certain proteins or cells onto the sensor, occurring very fast after implantation, might result in incorrect readings too. A second important aspect for all active implants is their biostability: an implant should remain functional after implantation; hence no body fluids should penetrate into the device causing corrosion and disabling the correct functionality of the device. In this section we will present an overview of the foreign body response to an implant, and strategies aimed to solve specific biocompatibility and biostability issues.

7.1. Biocompatibility

Biocompatibility has been defined as the ability of a material to perform with an appropriate host response in a specific application [207,208]. An appropriate host reaction includes no local tissue damage due to cytotoxicity, limited FBR and no short or long term adverse body effects. Also, this definition points out that a precise metric of biocompatibility does not exist. The site and duration of the implant, as well as type of material-body contact are important parameters, which makes biocompatibility a contextual concept. Even more, the same material can induce adverse or favourable biological response according to its particular employ [209].

Experiments to test biocompatibility are carried both *in vitro* and *in vivo*, in order to test the local and systemic effect of the materials. The ISO 10993 standard [210] is a well-known and constantly updated standard for evaluation of biocompatibility prior to clinical testing of a device. Various tests are essential for each material/device before medical use. The three most basic evaluations are cytotoxicity, sensitization, and irritation testing. The presence of toxic, leachable substances is investigated, causing adverse effects on biosystems. Based upon the final goal of an implant (implant duration, location in the body, and type of tissue contact), various additional categories of tests have to be performed, such as genotoxicity, hemocompatibility, carcinogenicity tests, *etc.* Obviously, a medical glove for short time contact with the skin will need much less testing compared to a heart assist device which will be implanted in the body for over 10 years.

7.2. The Foreign Body Response

In order to better understand how biocompatibility can be improved, it is useful to know how the body reacts to an foreign implant. A detailed description of this reaction can be found in [207,208]. Below a summary is given.

As introduced above, the natural response of the host to the introduction of exogenous material is called the *'foreign body response'* (FBR). Immediately after implantation in the body, the implant spontaneously acquires a layer of host proteins. Composition, concentration and conformation of this protein layer is dependent on the foreign material and the implantation site. As a consequence of the insertion, a fibrin clot is formed to close the lesion, while an oedema is formed close to the wound. Neutrophils and macrophages present at the injured site start to clean up the damaged area. These

leukocytes will release molecules, to signal to the immune system that a wound exists and to promote further infiltration of white cells at the injured site. In this phase, the host experiences a period of acute inflammation—the injured site will swell and is warm and painful.

Persistent inflammatory stimuli given by various leukocytes will lead to chronic inflammation. This stage is characterized by the presence of monocytes, macrophages, lymphocytes and plasma cells at the implant site, and a proliferation of blood vessels and connective tissue to heal the affected area. The macrophages will clean up the wound site, but they will also try to digest the foreign material. When the individual macrophages can't digest the implanted material/device, the macrophages form clusters called 'foreign body giant cells' in a final attempt to digest the foreign material. When using biocompatible materials, usually chronic inflammation last no longer than 2–3 weeks. In the final stage of the FBR, the body attempts to quarantine the external object in a fibrous capsule. Monocytes, macrophages and foreign body giant cells adhere at the implant interface. Fibrous tissue will form, surrounding these white blood cells, and walling the implant from the host tissue. Outside this fibrous tissue, a highly vascularized granulation tissue is formed.

7.3. Implant Design for Optimal Biocompatibility and Biostability

If biocompatibility issues are not addressed in a design, the device will have a disturbed functionality during or after the foreign body reaction, according to several factors:

- Grafting factors: depending on the implant surface morphology and properties, the FBR might be mild or strongly pronounced, resulting in a smooth integration in the local tissue or in the presence of a thick fibrous encapsulation or even a chronic infection.
- Implant fabrication factors: proper implant design should prevent adverse effects such as corrosion caused by biological fluids penetrating into the device, or release of toxic implant material in the organism.
- Sensor biofouling factors: due to accumulation of biological material on the sensor surface, the sensor performance might be impaired.

7.3.1. Grafting Factors

Upon implantation, the FBR should be mild, resulting in a smooth integration of the implant in the local tissue. The formation of a thick fibrous encapsulation during the FBR should be avoided in case the implant is carrying biosensors, since for most biosensors, this barrier will often result in an incorrect or a delayed sensor read-out. Finally, biofilm formation at the implant surface should be prevented at all means. Biofilm formation is the accumulation of bacteria at the implant surface, and is a cause of chronic infection. Due to the slimy substance used by the bacteria to encapsulate themselves, the removal of the bacterial colonies by antibiotics is almost impossible. Once a biofilm is formed, removal of the implant is often the only solution to overcome infection, hence prevention of biofilm formation by a fast and smooth integration of the implant into the local tissue is crucial.

Grafting solutions aim to improve cell adhesion, promote vascularization and reduce inflammatory stimuli. A first solution consists of the use of engineered implant coating or implant surface treatments. The material morphology, porosity and mechanical properties can influence cell signalling,

organization, proliferation and migration [211]. For example, at cellular dimensions, molecular diffusion is dictated by the geometry of the cellular interface with the material, the neighbouring cells, and the surrounding aqueous micro-environment. At larger scales (>10² μ m), material architecture determines bulk mechanical properties, cell migration, and nutrient and waste exchange [211]. In this respect, Sharkawy *et al.* [212] have shown that porous coatings of polyvinyl alcohol can promote higher vascular density, tissue in-growth and faster response to analytes compared to smooth surfaces. After this initial work, several groups have started to employ nanostructured shells for implantable sensors, using material like titanium [213] and silicon [214].

Other approaches, which promote blood vessel formation and reduce fibrosis, are based on porous poly-L-lactic acid coating [215], hydrogels [216–218], and coatings containing certain growth factors [219]. In order to reduce the inflammation and biofilm risk upon implantation, the use of drug releasing coatings can be considered, such as a coating of hydrogel loaded with dexamethasone, a strong anti-inflammatory agent [220,221].

7.3.2. Implant Fabrication Factors

Active implants typically consist of microchips, a battery, and other components, which are (at least partially) made from non-biocompatible materials. The diffusion of such materials into the body tissue should be avoided by all means. Furthermore, biofluids should not penetrate into an active implant: biofluids will corrode the device and might promote leaching of toxic compounds into the body. As a consequence, a bi-directional diffusion barrier is essential: body and implant must be isolated from each other. Moreover, biocompatible materials must be used for the fabrication of this diffusion barrier and the packaging of the total device [222,223]. The most conventional device package consists of a titanium box, equipped with feed-throughs for sensors. The titanium package of a pace-maker is well known, having dedicated ceramic/polymer-based feed-throughs for the leads. Titanium is biocompatible and the box is sufficiently thick to prevent all diffusion. Disadvantages of this package is its size and the rigidity of the box, which is most often in strong contrast with the local tissue and hence not ideal with respect to the resulting FBR. Smaller implant packages make use of encapsulation by ceramic materials and/or polymers.

In order to reduce implant dimensions, ceramic or polymer films can be used. They have the advantage of being compatible with wafer-level processing and to conform to any shape. Thin films can be divided into organic or inorganic.

Most popular organic compounds for packaging include SU-8, epoxies, silicones, polymides, polyurethanes and parylene-C. Attention should be paid when using these materials for long-term implants, since some of these materials can be affected by moisture penetration and biofluid etching. Among these compounds, parylene-C is very promising due to its excellent biocompatibility and step coverage, and its high resistance to moisture penetration [224–226]. A drawback of parylene-C is its limited resistance to heat. Polymide is very attractive as a flexible substrate for implantable electrodes [227]. Due to some remaining moisture permeability, additional measures should be taken to protect the implant from moisture penetration.

Among inorganic films, silicon nitride is the most widely employed for moisture protection [228]. Metal films can be also used for hermetic packaging, when deposited onto another polymeric film

which provides electrical isolation. Iridium oxide, titanium, titanium nitride, tantalum and tantalum nitride can be biocompatible if proper deposition conditions are respected [229]. Noble metals like platinum and gold present high corrosion resistance and biocompatibility, and represent excellent candidates for the microfabrication and the connection of the subcomponents of an implantable device. An example is the fabrication of a Utah Electrode Array as reported by Topper *et al.* [230]. Disadvantages of these metals are their ductility and the cost.

7.3.3. Biofouling Factors

Biofouling is the accumulation of biological material on the device surface. In contrast to biofilms, which consist of bacteria, this biologic material is not causing an infection. This aggregation of cells, macromolecules and small molecules on the biosensor membrane has been extensively reported as detrimental for the sensor function [231–234]. Biofouling often prevents diffusion of the analyte or adherence of the analyte on the sensor surface.

Several strategies have been employed to reduce biofouling, most of them rely on the use of dedicated membranes and coatings. An extensive review about this topic can be found in reference [235]. Below we will briefly describe two techniques for reducing biofouling: hydrogel overlays and Nafion coatings.

Hydrogel overlays, mainly made with poly(hydroxyethylmethacrylate) or poly(ethylene glycol), present a hydrophilic interface which can favor diffusion of water-soluble analytes. Diffusion rate is controlled by changing the crosslinking density of the gel. A drawback of this approach is their poor adhesion to the substrate, and a poor mechanical stability during the implant.

Nafion is a commercial, perfluorinated polymer that gained a lot of popularity as a coating for glucose sensors. It is characterized by chemical inertness and both hydrophobic and hydrophilic properties. An additional reason for its extensive use, is its easy application by dip-coating.

The natural response of the host to the introduction of exogenous material is called the 'foreign body response' (FBR). It basically consists in wound healing, combined with an attempt to digest followed by walling of the implant by fibrous tissue. Promoting smooth wound healing and minimizing the extent of the FBR is one of the major goals to accomplish when optimizing for biocompatibility of an implanted device. The selection of all components and raw materials for device fabrication (Figure 16) will influence its biocompatibility, as well as the integration techniques used to realize the final device.



Figure 16. A reliable biocompatible packaging.

Hence, biocompatibility is concept of primary importance, starting from the design of implantable biosensors, including all aspects of their fabrication, until the follow up of the implantation. This implies that in order to realize successful implantable devices, extensive communication between experts of different disciplines is essential starting from the most early development phase.

8. Security and Privacy

In order to complete a fully integrated sensing system, an end-to-end data acquisition system must be designed. This ranges from the control and calibration of the actual sensors, to analog-digital conversion of the data, to signal processing, local data storage, source coding, channel coding, transmission, and finally reliable, secure storage in back-end data-bases, and each of these needs to consider potential failure modes as well as threat models, while satisfying constraints on power and cost. Many of these techniques are well known and have been studied extensively in the field of sensor networks [236]. However, security and privacy are critical design aspects in biosensors due to their presence within and interactions with the human body.

Data security is typically characterized by three dimensions: (1) confidentiality; (2) integrity; and (3) availability [237]. Each of these presents unusual challenges in biosensors. As a foundational principle, all primary and derivative biometric data, including biosensor data originating from an individual should be considered private and owned by the user. Accordingly, data should be confidential and encrypted to avoid observation by unauthorized parties. If this privacy is violated, the data can be used in illegal or unethical ways, for example, insurance fraud or discrimination. The integrity of the data along with its origin and timestamp are also critical to preserve and protect. Data from a biosensor can be used to control drug-delivery systems [238] or electrical therapies [239], thus it is crucial to protect the data from tampering. Finally, the availability of the device should be protected such that it can perform its intended function, even in an environment remote from an administrator. Although many techniques can be applied from recent work in lightweight security and privacy for embedded devices such as wireless sensor nodes, RFID tags, and smart cards, biosensors present several different challenges.

8.1. Special Features for Biosensors

For this paper, we restrict ourselves to the class of biosensors, which actually perform as a lab on chip, conducting a small experiment at the molecular level on the sensor. Recent examples of this type include implantable/injectable subcutaneous devices that are remotely powered by a bandage-like patch which also provides a data-link to a higher level wearable device, possibly a body-area-network (BAN), and eventually higher levels of health information systems [164]. A related class of devices is that of low-cost disposable biosensors for detecting infectious disease or critical levels of glucose and lactate in a battlefield or other trauma situation [240]. However the constraints and threat models are significantly different since the device is not implanted. Data security and privacy issues for these types of sensors differ significantly from those of higher data-rate sensors such as brain [241] or intraocular [242] implants. For example:

- A particular sensor may take several minutes to complete its task, and only deliver several bytes of data. Thus, the information from the biosensor can have a very high value per bit. This presents challenges for cryptography since the space of possible data values is quite small.
- Energy issues range from remote-powered devices with very close power sources and energy harvesting to non-rechargeable batteries. Energy can be optimized by exploiting the very slow data rates, using sub-threshold voltage circuits or other techniques, which trade off performance for efficiency. However energy starvation is another threat vector [239].
- Error protection must be considered in addition to malicious tampering of data. Again, the very small amount of data means there is very little inherent redundancy, so error correction codes should be used. Integrity verification provides some built-in detection and correction of single bit errors but parity bits should be added to account for multiple bit errors [243].
- Threat scenarios are varied. Although eavesdropping on a properly operating sensor may be unlikely due to the very short wireless channel (a few mm for a sub-cutaneous sensor), impersonation of both reader and patch are plausible concerns. For example, the patch of an unconscious patient can be removed and replaced by another patch. Similarly, a rogue sensor can upload fraudulent data to a trusted patch.

Since biosensors create data, which is eventually used for larger scale health information systems and health care decisions, there are several additional considerations.

- Liability in health care may warrant the use of audit techniques. Sensor data can be used to help determine liability in case of malpractice or other wrong-doing [244].
- Security considers the impact of a malicious adversary. Safety issues related to operator error, design error or other unintentional malfunctions are probably a more likely concern and need to be balanced in the design.
- Emergency access to the sensor device may justify over-riding the security mechanism. A medical first-responder may need to disable the security of the device to gain access to life-critical data or to enable/disable a therapy (e.g., defibrillator) [239].
- Medical devices are regulated in most nations (e.g., FDA in the US) so it is important to detect data from unapproved devices.
- Medical devices are expensive and a prime target for counterfeiters. Cryptography techniques such as physical unclonable functions (PUFs) and watermarks can be used to identify genuine devices and the data they produce.
- Calibration, test and control of the sensor require a secure downlink. Threats arise due to deliberate miscalibration, which can result in erroneous measurements. Unsecured test and debug access is notorious for providing back-door access for attackers.

8.2. Alternative Architectures

Recently, a number of solutions have been presented to the general problem of implantable medical devices (IMDs) that include biosensors. In [245], a lightweight wireless protocol for IMDs is presented, leveraging existing technologies and emphasizing low-energy computation.

In [239], both eavesdropping and power-depletion attacks are considered. To address power-depletion, a novel but simple approach is developed that uses RFID-style remote powering until the authentication process is completed, at which point, more general access to the implanted device and battery are enabled.

The choice of encryption scheme should consider the nature of the data as well as the device constraints. In [246], the lightweight stream-cipher Hummingbird is implemented in hardware to encrypt data from a glucose sensor. In [242], block ciphers are explored for use in IMD security, although the target device is an ocular sensor with high data-rates.

In [247], a novel device worn around the neck provides a radio-level shield against unauthorized eavesdropping, jamming and access. This novel approach avoids modification of existing implanted devices (which would otherwise require new surgery to replace the device) and leverages the physical proximity of the shield, although it is external to the body.

In a similar approach, Sorber [248] proposes an amulet to provide broad protection for wearable medical devices. Cornelius [249] presents an intriguing approach based on accelerometers to detect if two devices are on the same body and thus experiencing similar movement. This can help prevent the challenges of the near-far problem where an eavesdropping or unauthorized device could actually be closer to the sensor than the intended receiver (e.g., in a crowded elevator). Document [250] proposes the use of keyed ultra-wideband communication in medical devices to allow low probability of intercept and even detection. UWB is also expected to provide better resistance to physical layer identification techniques based on RF signatures [251]. Biosensors present a set of new and diverse challenges for security and privacy and a unique combination of constraints. A recent review of these and the more general challenges of securing implantable medical devices can be found in [252].

9. Future Perspectives

Fully integrated biochip platforms for advanced healthcare are possible by considering all of the levels considered in this paper, from electro-chemistry, to biocompatibility, to sensor electronics, and eventually the security and privacy of the acquired data. Figure 17 shows a platform that can host two different micro-array biochips for two different applications. In the first case (on the right), several sensors having targets on drug compounds for anti-cancer treatments are on board. By choosing the right isoform of the probe enzymes from the family of cytochromes P450 we can develop a fully integrated platform for point-of-care application on chemotherapy patients. The considered drugs in the example are cyclophosphamide, etoposide, ftorafur, and torsemide, all compounds used in chemotherapy cocktails [253]. Of course, the possibility to monitor the amount of the anti-cancer compounds in the patients' blood provides the tuning and personalization of the therapy on patient's metabolism.

The example reported on the left of Figure 17 is instead related to the monitoring of both curing compounds and disease biomarkers. In fact, this configuration of the platform provides the measure of patient's metabolism biomarkers (glucose and ATP) as well as the sensing of anti-cancer (etoposide and cyclophosphamide) or anti-inflammatory (naproxen) drugs. As we have seen in Section 3, several other metabolites as target of the platform might be considered just by choosing the right probe enzymes to be immobilized onto the sensing micro-array.



Figure 17. Two possible integrated biochip platforms.





Platforms like that shown in Figure 17 require special co-design [254] of the supporting electronics circuitry for autonomous operations, as described in Section 5. So, a fully-integrated platform requires both an implanted device as well as a supporting external station for data receiving and power transmission, as schematically shown in Figure 18. The implantable device needs to have on board not only the detection front-end (Section 5) but also nano-materials to improve the sensors performance

(Section 4), biomaterials providing the right specificity (Section 3), may be a bit of microfluidics and a biocompatible package (Section 7) is necessary as well as the remote powering system (Section 8). The supporting external station has to be located not far away for constraints coming from the physics of remote powering. A good solution that has been discussed in Section 8 is to have an electronic patch as light as possible just on top of the skin over the implant region, as shown in Figure 19. In that manner, a mobile phone network assures geographical connection to the hospitals or the healthcare services.

Figure 19. The i-needle under the skin, the remote-powering patch, and the mobile phone for connection to a geographical network.



Gathering mobile network also means opening to remote control of patients in parallel from the hospital to elsewhere (Figure 20). So, full management of patients at home or during their normal life activities is then possible with such kind remote monitoring biochip platforms in order to assure advanced healthcare services for fast intervention in case of fatal events, personalized therapy, continuous support to chronic and elderly people, *etc*.

Figure 20. Applications to remote monitoring of human health.



10. Conclusions

This paper has presented an overview of the key issues to be addressed in developing fully integrated platforms for innovative biosensors with applications in advanced healthcare. We have seen how to improve the device sensitivity and to decrease the limit of detection by using different nanostructured materials (Section 2). We have discussed how to select the right bioreceptors and how to immobilize such bioreceptors (Section 3) on the sensing surfaces. We have considered the integration of bio and nano-materials onto micro-arrays (Section 4) by using the most advanced technology for heterogeneous systems. We have sketched appropriate CMOS circuits to provide the electronics front-end (Section 5) and the required remote powering (Section 6) to develop highly integrated implantable devices. We have highlighted the relevant problems of device biocompatibility (Section 7) and data security (Section 8) in order to show the future perspective of these fully integrated systems for advanced healthcare (Section 9). Full integration means solving all the problems concurrently and in common between the different sub-systems, and solving the new problems that usually arise from the integration of parts that are not automatically compatible—systems integration. Typical examples are the case of different bioprobes requiring different electrochemical queries facing a single common electronics front-end or that of the opposite needs in remote powering: definitely small receiving antennas in the implant to reduce patient pain and large receiving antennas to provide enough energy to the implants. So, substantial ongoing efforts are required and expected from the worldwide research and development community to succeed in fully operating autonomous and low-cost integrated platforms for advanced healthcare. Despite these challenges, we expect that in a five-ten year horizon we can anticipate reliable technology for the remote telemetry of the human metabolism.

Acknowledgments

The authors thank Davide Sacchetto for drawing Figure 3. These works has been partially supported by the i-IronIC project, financed with a grant from the Swiss Nano-Tera.ch initiative and evaluated by the Swiss National Science Foundation. The research has been also supported by the grants Sinergia CRSII2_127547/1 by Swiss National Science Foundation and PR023081/DAMD17-03-1-0172 by US Department of Defense (DoDPRMRP). The work on data security has been supported by the US Department of Health and Human Services.

References

- Murg, S.; Moore, B.; Lovell, M.; Wolf, R.; Bejjany, B.A.; Wu, A.H.B.; Kelles, S. *Business* Strategy for Molecular Diagnostics in the Lab, Report Summary, 2nd ed.; Washington G-2 Reports; Kennedy Information: Peterborough, NH, USA, 2009.
- 2. 10 Facts on Noncommunicable Diseases. Available online: http://www.who.int/features/factfiles/ noncommunicable_diseases/en/index.html (accessed on 18 June 2012).
- 3. Valgimigli, F.; Lucarelli, F.; Scuffi, C.; Morandi, S.; Sposato, I. Evaluating the clinical accuracy of GlucoMen®Day: A novel microdialysis-based continuous glucose monitor. *J. Diabetes Sci. Technol.* **2010**, *4*, 1182–1192.

- 4. Poscia, A.; Messeri, D.; Moscone, D.; Ricci, F.; Valgimigli, F. A novel continuous subcutaneous lactate monitoring system. *Biosen. Bioelectron.* **2005**, *20*, 2244–2250.
- Carrara, S.; Bolomey, L.; Boero, C.; Cavallini, A.; Meurville, E.; De Micheli, G.; Grassi, F.; Rezzonico, T. Single-Metabolite Bio-Nano-Sensors and System for Remote Monitoring in Animal Models. In *Proceedings of IEEE Sensors 2011 Conference*, Limerick, Ireland, 28–31 October 2011; pp. 716–719.
- 6. Cavallini, A.; De Micheli, G.; Carrara, S. Comparison of Three methods of biocompatible multi-walled carbon nanotubes confinement for the development of implantable amperometric ATP biosensors. *Sens. Lett.* **2011**, *9*, 1838–1844.
- 7. Carrara, S.; Cavallini, A.; Erokhin, V.; De Micheli, G. Multi-panel drugs detection in human serum for personalized therapy. *Biosen. Bioelectron.* **2011**, *26*, 3914–3919.
- Yu, B.; Long, N.; Moussy, Y.; Moussy, F. A long-term flexible minimally-invasive implantable glucose biosensor based on an epoxy-enhanced polyurethane membrane. *Biosen. Bioelectron.* 2006, *21*, 2275–2282.
- 9. Gough, D.A.; Kumosa, L.S.; Routh, T.L.; Lin, J.T.; Lucisano, J.Y. Function of an implanted tissue glucose sensor for more than 1 year in animals. *Sci. Trans. Med.* **2010**, *2*, 1–8.
- Carrara, S.; Shumyantseva, V.V.; Archakov, A.I.; Samorì, B. Screen-printed electrodes based on carbon nanotubes and cytochrome P450scc for highly sensitive cholesterol biosensors. *Biosen. Bioelectron.* 2008, 24, 148–150.
- Carrara, S.; Torre, M.D.; Cavallini, A.; De Venuto, D.; De Micheli, G. Multiplexing pH and Temperature in a Molecular Biosensor. In *Proceedings of the IEEE BioCAS 2010 Conference*, Paphos, Cyprus, 3–5 November 2010; pp. 146–149.
- 12. Forghanizadeh, H.P.; Rincon-Mora, G.A. Low power CMOS ramp generator circuit for DC-DC converters. *J. Low Power Electron.* **2006**, *2*, 1–5.
- 13. Olivo, J.; Carrara, S.; De Micheli, G. Energy harvesting and remote powering for implantable biosensors. *IEEE Sens. J.* **2011**, *11*, 1573–1586.
- 14. Grieshaber, D.; MacKenzie, R.; Vörös, J.; Reimhult, E. Electrochemical biosensors—Sensor principles and architectures. *Sensors* **2008**, *8*, 1400–1458.
- 15. Pavlickova, P.; Schneider, E.M.; Hug, H. Advances in recombinant antibody microarrays. *Clin. Chim. Acta* **2004**, *343*, 17–35.
- Song, Y.; Wang, L.; Ren, C.; Zhu, G.; Li, Z. A novel hydrogen peroxide sensor based on horseradish peroxidase immobilized in DNA films on a gold electrode. *Sens. Actuators B Chem.* 2006, 114, 1001–1006.
- 17. Compton, S.R.G.; Hallaj, R. Glucose biosensor prepared by glucose oxidase encapsulated sol-gel and carbon-nanotube-modified basal plane pyrolytic graphite electrode. *Anal. Biochem.* **2004**, *333*, 49–56.
- O'Neill, R.D.; Chang, S.-C.; Lowry, J.P.; McNeil, C.J. Comparisons of platinum, gold, palladium and glassy carbon as electrode materials in the design of biosensors for glutamate. *Biosen. Bioelectron.* 2004, 19, 1521–1528.
- 19. Patel, N.G.; Patel, P.D.; Vaishnav, V.S. Indium tin oxide (ITO) thin film gas sensor for detection of methanol at room temperature. *Sens. Actuators B Chem.* **2003**, *96*, 180–189.

- 20. Jan Krejcí, J.P.; Fujcik, L.; Khatib, S.; Hejátková, E.; Jakubka, L.; Giannoudi, L. Screen-printed sensors with graphite electrodes—Comparison of properties and physical method of sensitivity enhancement. *Microelectron. Int.* **2004**, *21*, 20–24.
- 21. Renedo, O.D.; Alonso-Lomillo, M.A.; Martínez, M.J.A. Recent developments in the field of screen-printed electrodes and their related applications. *Talanta* **2007**, *73*, 202–219.
- 22. Laschi, S.; Palchetti, I.; Mascini, M. Gold-based screen-printed sensor for detection of trace lead. *Sens. Actuators B Chem.* **2006**, *114*, 460–465.
- 23. Gerard, M.; Chaubey, A.; Malhotra, B.D. Application of conducting polymers to biosensors. *Biosen. Bioelectron.* **2002**, *17*, 345–359.
- Guiseppi-Elie, A.; Brahim S.; Wilson, A. Biosensors Based on Electrically Conducting Polymers. In *Handbook of Conducting Polymers: Conjugated Polymer Processing and Applications*, 3rd ed.; Skotheim, T., Reynolds, J.R., Eds.; Taylor and Francis: New York, NY, USA, 2007; Chapter 12, pp. 1–45.
- 25. Kerman, K.; Saito, M.; Tamiya, E.; Yamamura, S.; Takamura, Y. Nanomaterial-based electrochemical biosensors for medical applications. *TrAC Trend. Anal. Chem.* **2008**, *27*, 585–592.
- 26. Yogeswaran, U.; Chen, S.-M. A review on the electrochemical sensors and biosensors composed of nanowires as sensing material. *Sensors* **2008**, *8*, 290–313.
- 27. Agüí, L.; Yáñez-Sedeño, P.; Pingarrón, J.M. Role of carbon nanotubes in electroanalytical chemistry: A review. *Anal. Chim. Acta* **2008**, *622*, 11–47.
- Pumera, M.; Sánchez, S.; Ichinose, I.; Tang, J. Electrochemical nanobiosensors. *Sens. Actuators B Chem.* 2007, *123*, 1195–1205.
- 29. Carrara, S. Nano-bio-technology and sensing chips: New electrochemical systems for detection in personalized therapies and cell biology. *Sensors* **2010**, *10*, 526–543.
- 30. Habermüller, K.; Mosbach, M.; Schuhmann, W. Electron-transfer mechanisms in amperometric biosensors. *Fresenius' J. Anal. Chem.* **2000**, *366*, 560–568.
- 31. Guiseppi-Elie, A.; Lei, C.; Baughman, R.H. Direct electron transfer to glucose oxidase using carbon nanotubes. *Nanotechnology* **2002**, *13*, 559–564.
- 32. Wang, J. Carbon-nanotube based electrochemical biosensors: A review. *Electroanalysis* 2005, 17, 7–14.
- 33. Zhao, Q.; Gan, Z.; Zhuang, Q. Electrochemical sensors based on carbon nanotubes. *Electroanalysis* **2002**, *14*, 1609–1613.
- 34. Lin, Y.; Lu, F.; Tu, Y.; Ren, Z. Glucose biosensors based on carbon nanotube nanoelectrode ensembles. *Nano Lett.* **2003**, *4*, 191–195.
- 35. Gooding, J.J. Nanostructuring electrodes with carbon nanotubes: A review on electrochemistry and applications for sensing. *Electrochim. Acta* **2005**, *50*, 3049–3060.
- 36. Merkoçi, C. Carbon nanotubes in analytical sciences. *Microchim. Acta* 2006, 152, 157–174.
- 37. Ahammad, J.S.; Lee, J.-J.; Rahman, M.A. Electrochemical sensors based on carbon nanotubes. *Sensors* **2009**, *9*, 2289–2319.
- 38. Yang, F.W.; Chen, X.; Evans, D.G. Direct electrochemistry of glucose oxidase on a graphite nanosheet–Nafion composite film modified electrode. *Electrochem. Commun.* **2009**, *11*, 997–1000.

- Park, C.; Ounaies, Z.; Watson, K.A.; Crooks, R.E.; Smith, J., Jr.; Lowther, S.E.; Connell, J.W.; Siochi, E.J.; Harrison, J.S.; Clair, T.L.S. Dispersion of single wall carbon nanotubes by *in situ* polymerization under sonication. *Chem. Phys. Lett.* 2002, *364*, 303–308.
- 40. Ghica, M.E.; Pauliukaite, R.; Fatibello-Filho, O.; Brett, C.M.A. Application of functionalised carbon nanotubes immobilised into chitosan films in amperometric enzyme biosensors. *Sens. Actuators B Chem.* **2009**, *142*, 308–315.
- 41. Mohammadi, A.H.; Bourais, I.; Palleschi, G. Enzyme inhibition-based biosensors for food safety and environmental monitoring. *Biosens. Bioelectron.* **2006**, *21*, 1405–1423.
- 42. Shimoda, H.; Oh, S.J.; Geng, H.Z.; Walker, R.J.; Zhang, X.B.; McNeil, L.E.; Zhou, O. Self-assembly of carbon nanotubes. *Adv. Mater.* **2002**, *14*, 899–901.
- Shumyantseva, V.V.; Carrara, S.; Bavastrello, V.; Riley, J.D.; Bulko, T.V.; Usanov, S.A.; Nicolini, C.; Archakov, A.I. Direct electron transfer between cytochrome P450scc and screenprinted rhodium graphite electrode modified with gold nanoparticles. *Biosens. Bioelectron.* 2005, 21, 217–222.
- Carrara, S.; Bavastrello, V.; Ricci, D.; Stura, E.; Nicolini, C. Improved nanocomposite materials for biosensor applications investigated by Impedance Spectroscopy. *Sens. Actuators B Chem.* 2005, 109, 221–226.
- 45. Georgakilas, V.; Gournis, D.; Tzitzios, V.; Pasquato, L.; Guldi, D.M.; Prato, M. Decorating carbon nanotubes with metal or semiconductor nanoparticles. *J. Mater. Chem.* **2007**, *17*, 2679–2694.
- 46. Wildgoose, G.G.; Banks, C.E.; Compton, R.G. Metal nanoparticles and related materials supported on carbon nanotubes: Methods and applications. *Small* **2006**, *2*, 182–193.
- Yang, M.; Yang, Y.; Liu, Y.; Shen, G.; Yu, R. Platinum nanoparticles-doped sol-gel/carbon nanotubes composite electrochemical sensors and biosensors. *Biosens. Bioelectron.* 2006, 21, 1125–1131.
- 48. Gopalan, R.A.I.; Lee, K.-P. Synergistic contributions of multiwall carbon nanotubes and gold nanoparticles in a chitosan–ionic liquid matrix towards improved performance for a glucose sensor. *Electrochem. Commun.* **2009**, *11*, 397–401.
- Cui, H.-F.; Ye, J.-S.; Liu, X.; Zhang, W.-D.; Sheu, F.-S. Pt-Pb alloy nanoparticle/carbon nanotube nanocomposite: A strong electrocatalyst for glucose oxidation. *Nanotechnology* 2006, 17, 2334–2339.
- Zhao, Y.; Y.E.; Fan, L.; Qiu, Y.; Yang, S. A new route for the electrodeposition of platinumnickel alloy nanoparticles on multi-walled carbon nanotubes. *Electrochim. Acta* 2007, *52*, 5873–5878.
- Zhao, Y.; Fan, L.; Zhong, H.; Li, Y.; Yang, S. Platinum nanoparticle clusters immobilized on multiwalled carbon nanotubes: Electrodeposition and enhanced electrocatalytic activity for methanol oxidation. *Adv. Funct. Mater.* 2007, *17*, 1537–1541.
- Lu, G.; Chen, J. Controlled decoration of carbon nanotubes with nanoparticles. *Nanotechnology* 2006, *17*, 2891–2896.
- 53. Hou, Y.; Cheng, Y.; Hobson, T.; Liu, J. Design and synthesis of hierarchical MnO₂ nanospheres/carbon nanotubes/conducting polymer ternary composite for high performance electrochemical electrodes. *Nano Lett.* **2010**, *10*, 2727–2733.

- Baibarac M.; mez-Romero, P. Nanocomposites based on conducting polymers and carbon nanotubes: From fancy materials to functional applications. J. Nanosci. Nanotechnol. 2006, 6, 289–302.
- 55. Reddy, K.R.; Sin, B.C.; Ryu, K.S.; Kim, J.-C.; Chung, H.; Lee, Y. Conducting polymer functionalized multi-walled carbon nanotubes with noble metal nanoparticles: Synthesis, morphological characteristics and electrical properties. *Synth. Met.* **2009**, *159*, 595–603.
- 56. Lange, U.; Roznyatovskaya, N.V.; Mirsky, V.M. Conducting polymers in chemical sensors and arrays. *Anal. Chim. Acta* **2008**, *614*, 1–26.
- 57. Malhotra, B.D.; Chaubey, A.; Singh, S.P. Prospects of conducting polymers in biosensors. *Anal. Chim. Acta* **2006**, *578*, 59–74.
- Cosnier, S. Biosensors based on electropolymerized films: New trends. *Anal. Bioanal. Chem.* 2003, 377, 507–520.
- 59. Guiseppi-Elie, A. Electroconductive hydrogels: Synthesis, characterization and biomedical applications. *Biomaterials* **2010**, *31*, 2701–2716.
- Li, Y.; Wang, P.; Wang, L.; Lin, X. Overoxidized polypyrrole film directed single-walled carbon nanotubes immobilization on glassy carbon electrode and its sensing applications. *Biosens. Bioelectron.* 2007, 22, 3120–3125.
- Yuan, W.S.; Li, G.L.; Neoh, K.G.; Kang, E.T. Glucose biosensor from covalent immobilization of chitosan-coupled carbon nanotubes on polyaniline-modified gold electrode. *ACS Appl. Mater. Interface* 2010, *2*, 3083–3091.
- Tsai, Y.-C.; Huang, J.-D.; Chiu, C.-C. Amperometric ethanol biosensor based on poly(vinyl alcohol)–multiwalled carbon nanotube-alcohol dehydrogenase biocomposite. *Biosens. Bioelectron.* 2007, *22*, 3051–3056.
- Wang, Z.; Xiao, S.; Chen Y. β-Cyclodextrin incorporated carbon nanotubes-modified electrodes for simultaneous determination of adenine and guanine. J. Electroanal. Chem. 2006, 589, 237–242.
- Xiao, F.; Ruan, C.; Liu, L.; Yan, R.; Zhao, F.; Zeng, B. Single-walled carbon nanotube-ionic liquid paste electrode for the sensitive voltammetric determination of folic acid. *Sens. Actuators B Chem.* 2008, 134, 895–901.
- Tan, X.; Li, M.; Cai, P.; Luo, L.; Zou, X. An amperometric cholesterol biosensor based on multiwalled carbon nanotubes and organically modified sol-gel/chitosan hybrid composite film. *Anal. Biochem.* 2005, 337, 111–120.
- 66. Kang, X.; Mai, Z.; Zou, X.; Cai, P.; Mo, J. Glucose biosensors based on platinum nanoparticlesdeposited carbon nanotubes in sol–gel chitosan/silica hybrid. *Talanta* **2008**, *74*, 879–886.
- 67. Zhu, L.; Tian, C.; Zhai, J.; Yang, R. Sol-gel derived carbon nanotubes ceramic composite electrodes for electrochemical sensing. *Sens. Actuators B Chem.* **2007**, *125*, 254–261.
- Steckel, J.S.; Snee, P.; Coe-Sullivan, S.; Zimmer, J.P.; Halpert, J.E.; Anikeeva, P.; Kim, L.-A.; Bulovic, V.; Bawendi, M.G. Color-saturated green-emitting QD-LEDs. *Angew. Chem. Int. Edn.* 2006, 45, 5796–5800.
- 69. Algar, W.R.; Tavares, A.J.; Krull, U.J. Beyond labels: A review of the application of quantum dots as integrated components of assays, bioprobes, and biosensors utilizing optical transduction. *Anal. Chim. Acta* **2010**, *673*, 1–25.

- 70. Niemeyer, C.M. Functional hybrid devices of proteins and inorganic nanoparticles. *Angew. Chem. Int. Edn.* **2003**, *42*, 5796–5800.
- 71. Bönnemann, H.; Richards, R.M. Nanoscopic metal particles—Synthetic methods and potential applications. *Eur. J. Inorg. Chem.* **2001**, *2001*, 2455–2480.
- 72. Iost, R.M.; Madurro, J.M.; Brito-Madurro, A.G.; Nantes, I.L.; Caseli, L.; Crespilho, F.N. Strategies of nano-manipulation for application in electrochemical biosensors. *Int. J. Electrochem. Sci.* 2011, *6*, 2965–2997.
- Shukla, R.; Bansal, V.; Chaudhary, M.; Basu, A.; Bhonde, R.R.; Sastry, M. Biocompatibility of gold nanoparticles and their endocytotic fate inside the cellular compartment: A microscopic overview. *Langmuir* 2005, *21*, 10644–10654.
- 74. Wanekaya, K.; Chen, W.; Myung, N.V.; Mulchandani, A. Nanowire-based electrochemical biosensors. *Electroanalysis* **2006**, *18*, 533–550.
- 75. Heidelberg, W.A.; Boland, J.J. Mechanical properties of ultrahigh-strength gold nanowires. *Nat. Mater.* **2005**, *4*, 525–529.
- 76. Teng, X.; Han, W.-Q.; Ku, W.; Hücker, M. Synthesis of ultrathin palladium and platinum nanowires and a study of their magnetic properties. *Angew. Chem.* **2008**, *120*, 2085–2088.
- Fang, Y.-P.; Xu, A.-W.; Song, R.-Q.; Zhang, H.-X.; You, L.-P.; Yu, J.C.; Liu, H.-Q. Systematic synthesis and characterization of single-crystal lanthanide orthophosphate nanowires. *J. Am. Chem. Soc.* 2003, *125*, 16025–16034.
- 78. Aihua, L. Towards development of chemosensors and biosensors with metal-oxide-based nanowires or nanotubes. *Biosens. Bioelectron.* **2008**, *24*, 167–177.
- 79. Schmidt, V.; Wittemann, J.V.; Senz, S.; Gösele, U. Silicon nanowires: A review on aspects of their growth and their electrical properties. *Adv. Mater.* **2009**, *21*, 2681–2702.
- 80. Shao, Y.; Wang, J.; Wu, H.; Liu, J.; Aksay, I.A.; Lin, Y. Graphene based electrochemical sensors And biosensors: A review. *Electroanalysis* **2010**, *22*, 1027–1036.
- 81. Zhou, M.; Zhai, Y.; Dong, S. Electrochemical sensing and biosensing platform based on chemically reduced graphene oxide. *Anal. Chem.* **2009**, *81*, 5603–5613.
- Shan, C.; Yang, H.; Han, D.; Zhang, Q.; Ivaska, A.; Niu, L. Electrochemical determination of NADH and ethanol based on ionic liquid-functionalized graphene. *Biosens. Bioelectron.* 2010, 25, 1504–1508.
- 83. Shan, C.; Yang, H.; Han, D.; Zhang, Q.; Ivaska, A.; Niu, L. Graphene/AuNPs/chitosan nanocomposites film for glucose biosensing. *Biosens. Bioelectron.* **2010**, *25*, 1070–1074.
- Wang, Z.; Zhou, X.; Zhang, J.; Boey, F.; Zhang, H. Direct electrochemical reduction of single-layer graphene oxide and subsequent functionalization with glucose oxidase. *J. Phys. Chem. C* 2009, *113*, 14071–14075.
- 85. Sotiropoulou, S.; Gavalas, V.; Vamvakaki, V.; Chaniotakis, N.A. Novel carbon materials in biosensor systems. *Biosens. Bioelectron.* **2003**, *18*, 211–215.
- 86. Rubianes M.D.; Rivas, G.A. Carbon nanotubes paste electrode. *Electrochem. Commun.* **2003**, *5*, 689–694.
- 87. Wu, L.; Zhang, X.; Ju, H. Amperometric glucose sensor based on catalytic reduction of dissolved oxygen at soluble carbon nanofiber. *Biosens. Bioelectron.* **2007**, *23*, 479–484.

- Lu, J.; Drzal, L.T.; Worden, R.M.; Lee, I. Simple fabrication of a highly sensitive glucose biosensor using enzymes immobilized in exfoliated graphite nanoplatelets nafion membrane. *Chem. Mater.* 2007, 19, 6240–6246.
- 89. Zhou, M.; Shang, L.; Li, B.; Huang, L.; Dong, S. Highly ordered mesoporous carbons as electrode material for the construction of electrochemical dehydrogenase- and oxidase-based biosensors. *Biosens. Bioelectron.* **2008**, *24*, 442–447.
- 90. Ahuja, R.T.; Kumar, D. Recent progress in the development of nano-structured conducting polymers/nanocomposites for sensor applications. *Sens. Actuators B Chem.* **2009**, *136*, 275–286.
- Wanekaya, K.; Bangar, M.A.; Yun, M.; Chen, W.; Myung, N.V.; Mulchandani, A. Field-effect transistors based on single nanowires of conducting polymers. J. Phys. Chem. C 2007, 111, 5218–5221.
- 92. Balazs, C.; Emrick, T.; Russell, T.P. Nanoparticle polymer composites: Where two small worlds meet. *Science* **2006**, *314*, 1107–1110.
- 93. Leca-Bouvier, D.; Blum, L.J. *Enzyme for Biosensing Applications Recognition Receptors in Biosensors*; Zourob, M., Ed.; Springer: New York, NY, USA, 2010; pp. 177–220.
- Logrieco, A.; Arrigan, D.W.; Brengel-Pesce, K.; Siciliano, P.; Tothill, I. DNA arrays, electronic noses and tongues, biosensors and receptors for rapid detection of toxigenic fungi and mycotoxins: A review. *Food Addit. Contam.* 2005, *22*, 335–344.
- 95. Ricci, F.; Volpe, G.; Micheli, L.; Palleschi, G. A review on novel developments and applications of immunosensors in food analysis. *Anal. Chim. Acta* **2007**, *605*, 111–129.
- 96. Zhang, L.; Peng, H.; Kilmartin, P.A.; Soeller, C.; Travas-Sejdic, J. Polymeric acid doped polyaniline nanotubes for oligonucleotide sensors. *Electroanalysis* **2007**, *19*, 870–875.
- 97. Lu, Y.; Li, X.; Zhang, L.; Yu, P.; Su, L.; Mao, L. Aptamer-based electrochemical sensors with aptamer-complementary DNA oligonucleotides as probe. *Anal. Chem.* **2008**, *80*, 1883–1890.
- 98. Kumar S.; Chen, S.-M. Electroanalysis of NADH using conducting and redox active polymer/carbon nanotubes modified electrodes-a review. *Sensors* **2008**, *8*, 739–766.
- 99. Shu, W.; Laue, E.D.; Seshia, A.A. Investigation of biotin–streptavidin binding interactions using microcantilever sensors. *Biosens. Bioelectron.* **2007**, *22*, 2003–2009.
- 100. Katz, E.; Willner, B. Integration of layered redox proteins and conductive supports for bioelectronic applications. *Angew. Chem. Int. Edn.* **2000**, *39*, 1180–1218.
- 101. Katz, W.E.; Willner, B. Electrical contact of redox enzyme layers associated with electrodes: Routes to amperometric biosensors. *Electroanalysis* **1997**, *9*, 965–977.
- 102. Oliver, N.S.; Toumazou, C.; Cass, A.E.G.; Johnston, D.G. Glucose sensors: A review of current and emerging technology. *Diabet. Med.* **2009**, *26*, 197–210.
- 103. Galarneau, A.; Primeau, M.; Trudeau, L.-E.; Michnick, S.W. [beta]-Lactamase protein fragment complementation assays as *in vivo* and *in vitro* sensors of protein-protein interactions. *Nat. Biotechnol.* 2002, 20, 619–622.
- Herlem, L.G.; Lakard, S.; Antoniou, A.; Fahys, B. Urea potentiometric biosensor based on modified electrodes with urease immobilized on polyethylenimine films. *Biosens. Bioelectron.* 2004, 19, 1641–1647.
- 105. Wang, J. Amperometric biosensors for clinical and therapeutic drug monitoring: A review. *J. Pharm. Biomed. Anal.* **1999**, *19*, 47–53.

- 106. Guengerich, F.P. Cytochrome P450 enzymes in the generation of commercial products. *Nat. Rev. Drug Discov.* **2002**, *1*, 359–366.
- 107. Guengerich, F.P. Cytochrome p450 and chemical toxicology. *Chem. Res. Toxicol.* 2008, 21, 70-83.
- 108. Johnson, L.; Lewis, B.C.; Elliot, D.J.; Miners, J.O.; Martin, L.L. Electrochemical characterisation of the human cytochrome P450 CYP2C9. *Biochem. Pharmacol.* **2005**, *69*, 1533–1541.
- 109. Bistolas, N.; Wollenberger, U.; Jung, C.; Scheller, F.W. Cytochrome P450 biosensors—A review. *Biosens. Bioelectron.* 2005, 20, 2408–2423.
- 110. Sharma, S.K.; Sehgal, N.; Kumar, A. Biomolecules for development of biosensors and their applications. *Curr. Appl. Phys.* **2003**, *3*, 307–316.
- 111. Ahuja, T.; Mir, I.A.; Kumar, D.; Rajesh, R. Biomolecular immobilization on conducting polymers for biosensing applications. *Biomaterials* **2007**, *28*, 791–805.
- 112. Rusmini, F.; Zhong, Z.; Feijen, J. Protein immobilization strategies for protein biochips. *Biomacromolecules* 2007, *8*, 1775–1789.
- 113. Joseph, S.; Rusling, J.F.; Lvov, Y.M.; Friedberg, T.; Fuhr, U. An amperometric biosensor with human CYP3A4 as a novel drug screening tool. *Biochem. Pharmacol.* **2003**, *65*, 1817–1826.
- 114. Tkac, J.; Ruzgas, T. Dispersion of single walled carbon nanotubes. Comparison of different dispersing strategies for preparation of modified electrodes toward hydrogen peroxide detection. *Electrochem. Commun.* 2006, *8*, 899–903.
- 115. Kang, X.; Mai, Z.; Zou, X.; Cai, P.; Mo, J. A novel glucose biosensor based on immobilization of glucose oxidase in chitosan on a glassy carbon electrode modified with gold–platinum alloy nanoparticles/multiwall carbon nanotubes. *Anal. Biochem.* **2007**, *369*, 71–79.
- 116. Qian, L.; Yang, X. Composite film of carbon nanotubes and chitosan for preparation of amperometric hydrogen peroxide biosensor. *Talanta* **2006**, *68*, 721–727.
- 117. Yi, H.; Wu, L.-Q.; Bentley, W.E.; Ghodssi, R.; Rubloff, G.W.; Culver, J.N.; Payne, G.F. Biofabrication with chitosan. *Biomacromolecules* **2005**, *6*, 2881–2894.
- 118. He, C.; Lin, J.; Zhao, Y.; Zhang, S. One-step synthesis of silver nanoparticles/carbon nanotubes/chitosan film and its application in glucose biosensor. *Sens. Actuators B Chem.* 2009, 137, 768–773.
- 119. Zhou, Q.; Xie, Q.; Fu, Y.; Su, Z.; Jia, X.; Yao, S. Electrodeposition of carbon nanotubeschitosan-glucose oxidase biosensing composite films triggered by reduction of p-benzoquinone or H₂O₂. J. Phys. Chem. B 2007, 111, 11276–11284.
- Luo, X.-L.; Xu, J.-J.; Du, Y.; Chen, H.-Y. A glucose biosensor based on chitosan–glucose oxidase–gold nanoparticles biocomposite formed by one-step electrodeposition. *Anal. Biochem.* 2004, *334*, 284–289.
- Tsai Y.-C.; Chiu, C.-C. Amperometric biosensors based on multiwalled carbon nanotube-Nafiontyrosinase nanobiocomposites for the determination of phenolic compounds. *Sens. Actuators B Chem.* 2007, 125, 10–16.
- 122. Musameh, W.M.; Lin, Y. Solubilization of carbon nanotubes by nafion toward the preparation of amperometric biosensors. J. Am. Chem. Soc. 2003, 125, 2408–2409.

- 123. Huang, K.-J.; Liu, X.; Xie, W.-Z.; Yuan, H.-X. Electrochemical behavior and voltammetric determination of norfloxacin at glassy carbon electrode modified with multi walled carbon nanotubes/Nafion. *Colloids Surf. B Biointerfaces* **2008**, *64*, 269–274.
- 124. Andreescu, B.S.; Marty, J.L. Affinity methods to immobilize acetylcholinesterases for manufacturing biosensors. *Anal. Lett.* **2004**, *37*, 1571–1588.
- 125. Chaubey, R.; Malhotra, B.D. Mediated biosensors. Biosens. Bioelectron. 2002, 17, 441-456.
- 126. Gambhir, M.G.; Mulchandani, A.; Malhotra, B. Coimmobilization of urease and glutamate dehydrogenase in electrochemically prepared polypyrrole-polyvinyl sulfonate films. *Appl. Biochem. Biotechnol.* 2001, *96*, 249–257.
- 127. Geetha, S.; Rao, C.R.K.; Vijayan, M.; Trivedi, D.C. Biosensing and drug delivery by polypyrrole, *Anal. Chim. Acta* **2006**, *568*, 119–125.
- 128. Vidal, J.-C.; Garcia-Ruiz, E.; Castillo, J.-R. Recent advances in electropolymerized conducting polymers in amperometric biosensors. *Microchim. Acta* **2003**, *143*, 93–111.
- 129. Rahman, M.A.; Park, D.-S.; Chang, S.-C.; McNeil, C.J.; Shim, Y.-B. The biosensor based on the pyruvate oxidase modified conducting polymer for phosphate ions determinations. *Biosens. Bioelectron.* **2006**, *21*, 1116–1124.
- 130. Xu, Z.; Chen, X.; Dong, S. Electrochemical biosensors based on advanced bioimmobilization matrices. *TrAC Trends Anal. Chem.* **2006**, *25*, 899–908.
- 131. López-Cabarcos, R.-R.E.; López-Ruiz, B. High stability amperometric biosensor based on enzyme entrapment in microgels. *Talanta* **2005**, *68*, 99–107.
- 132. Barriet, G.D.; Sletmoen, M.; Stokke, B.T. Responsive hydrogels for label-free signal transduction within biosensors. *Sensors* **2010**, *10*, 4381–4409.
- 133. Ulijn, R.V.; Bibi, N.; Jayawarna, V.; Thornton, P.D.; Todd, S.J.; Mart, R.J.; Smith, A.M.; Gough, J.E. Bioresponsive hydrogels. *Mater. Today* **2007**, *10*, 40–48.
- 134. Gupta, R.; Chaudhury, N.K. Entrapment of biomolecules in sol-gel matrix for applications in biosensors: Problems and future prospects. *Biosens. Bioelectron.* **2007**, *22*, 2387–2399.
- 135. Chaubey, P.K.A.; Malhotra, B.D. Application of polyaniline/sol-gel derived tetraethylorthosilicate films to an amperometric lactate biosensor. *Anal. Sci.* **2003**, *19*, 1477–1480.
- 136. Fournier, B.D.; Danet, A.; Marty, J.-L. Biosensors based on highly sensitive acetylcholinesterases for enhanced carbamate insecticides detection. *Anal. Chim. Acta* 2006, 562, 115–121.
- 137. Giess, F.; Friedrich, M.G.; Heberle, J.; Naumann, R.L.; Knoll, W. The protein-tethered lipid bilayer: A novel mimic of the biological membrane. *Biophys. J.* **2004**, *87*, 3213–3220.
- 138. Jeuken, J.C.; Connell, S.D.; Henderson, P.J.F.; Gennis, R.B.; Evans, S.D.; Bushby, R.J. Redox enzymes in tethered membranes. *J. Am. Chem. Soc.* **2006**, *128*, 1711–1716.
- 139. Lojou, É.; Bianco, P. Membrane electrodes for protein and enzyme electrochemistry. *Electroanalysis* **2004**, *16*, 1113–1121.
- Mousty, F. Biosensing applications of clay-modified electrodes: A review. *Anal. Bioanal. Chem.* 2010, 396, 315–325.
- 141. Gascón, G.I.; Giménez, R.; Cea, P.; López, M.C.; Lafuente, C. Supramolecular architecture in langmuir films of a luminescent ionic liquid crystal. *J. Phys. Chem. C* **2009**, *113*, 18827–18834.

- 142. Santos, T.C.F.; Péres, L.O.; Wang, S.H.; Oliveira, O.N.; Caseli, L. Mixing alternating copolymers containing fluorenyl groups with phospholipids to obtain langmuir and langmuir–blodgett films. *Langmuir* **2009**, *26*, 5869–5875.
- 143. Tiemeyer, S.; Paulus, M.; Tolan, M. Effect of surface charge distribution on the adsorption orientation of proteins to lipid monolayers. *Langmuir* **2010**, *26*, 14064–14067.
- 144. Montanha, A.; Pavinatto, F.J.; Caseli, L.; Kaczmarek, O.; Liebscher, J.; Huster, D.; Oliveira, O.N. Properties of lipophilic nucleoside monolayers at the air–water interface. *Colloids Surf B Biointerfaces* 2010, 77, 161–165.
- 145. Pavinatto, F.J.; Pacholatti, C.P.; Montanha, E.A.; Caseli, L.; Silva, H.S.; Miranda, P.B.; Viitala, T.; Oliveira, O.N., Jr. Cholesterol mediates chitosan activity on phospholipid Monolayers and langmuir–blodgett films. *Langmuir* 2009, 25, 10051–10061.
- 146. Girard-Egrot, P.; Godoy, S.; Blum, L.J. Enzyme association with lipidic Langmuir–Blodgett films: Interests and applications in nanobioscience. *Adv. Colloid Interface Sci.* 2005, *116*, 205–225.
- 147. Schmidt, T.F.; Caseli, L.; Viitala, T.; Oliveira, O.N. Enhanced activity of horseradish peroxidase in Langmuir–Blodgett films of phospholipids. *Biochim. Biophys. Acta (BBA) Biomembr.* 2008, 1778, 2291–2297.
- 148. Kafi, F.Y.A.K.M.; Shin, H.-K.; Kwon, Y.-S. A novel amperometric hydrogen peroxide biosensor based on immobilization of hemoglobin in linoleic acid monolayer by Langmuir–Blodgett technique. *Colloids Surf. A Physicochem. Eng. Asp.* 2006, 284–285, 125–129.
- 149. Saiki, O.T.; Kusakari, A.; Endo, H.; Ichihara, M.; Izumi, M. Incorporation of glucose oxidase into langmuir-blodgett films based on prussian blue applied to amperometric glucose biosensor. *Langmuir* 2007, 23, 4675–4681.
- Choi, J.-W.; Kim, Y.-K.; Lee, I.-H.; Min, J.; Lee, W.H. Optical organophosphorus biosensor consisting of acetylcholinesterase/viologen hetero Langmuir-Blodgett film. *Biosens. Bioelectron.* 2001, *16*, 937–943.
- 151. Singhal, R.; Gambhir, A.; Pandey, M.K.; Annapoorni, S.; Malhotra, B.D. Immobilization of urease on poly(N-vinyl carbazole)/stearic acid Langmuir–Blodgett films for application to urea biosensor. *Biosens. Bioelectron.* 2002, 17, 697–703.
- 152. Pavinatto, F.J.; Fernandes, E.G.R.; Alessio, P.; Constantino, C.J.L.; de Saja, J.A.; Zucolotto, V.; Apetrei, C.; Oliveira, O.N; Rodriguez-Mendez, M.L. Optimized architecture for Tyrosinasecontaining Langmuir-Blodgett films to detect pyrogallol. *J. Mater. Chem.* 2011, *21*, 4995–5003.
- Girard-Egrot, P.; Morélis, R.M.; Coulet, P.R. Direct bioelectrochemical monitoring of choline oxidase kinetic behaviour in Langmuir–Blodgett nanostructures. *Bioelectrochem. Bioenerg.* 1998, 46, 39–44.
- 154. Sharma, S.K.; Singhal, R.; Malhotra, B.D.; Sehgal, N.; Kumar, A. Lactose biosensor based on Langmuir–Blodgett films of poly(3-hexyl thiophene). *Biosens. Bioelectron.* **2004**, *20*, 651–657.
- 155. Wang, X.; Yin, F.; Tu, Y. A uric acid biosensor based on Langmuir-Blodgett film as an enzymeimmobilizing matrix. *Anal. Lett.* **2010**, *43*, 1507–1515.
- 156. Siqueira, J.R., Jr.; Caseli, L.; Crespilho, F.N.; Zucolotto, V.; Oliveira, O.N. Immobilization of biomolecules on nanostructured films for biosensing. *Biosens. Bioelectron.* **2010**, *25*, 1254–1263.

- 157. Zhao, W.; Xu, J.-J.; Chen, H.-Y. Electrochemical biosensors based on layer-by-layer assemblies. *Electroanalysis* **2006**, *18*, 1737–1748.
- 158. Campàs, M.; O'Sullivan, C. Layer-by-layer biomolecular assemblies for enzyme sensors, immunosensing, and nanoarchitectures. *Anal. Lett.* **2003**, *36*, 2551–2569.
- Zou, X.Y.; Sun, L.-X.; Xu, F. Direct electron transfer of cytochrome c and its biosensor based on gold nanoparticles/room temperature ionic liquid/carbon nanotubes composite film. *Electrochem. Commun.* 2008, 10, 38–41.
- 160. Li, W.; Wang, Z.; Sun, C.; Xian, M.; Zhao, M. Fabrication of multilayer films containing horseradish peroxidase and polycation-bearing Os complex by means of electrostatic layer-by-layer adsorption and its application as a hydrogen peroxide sensor. *Anal. Chim. Acta* **2000**, *418*, 225–232.
- Ram, M.K.; Bertoncello, P.; Ding, H.; Paddeu, S.; Nicolini, C. Cholesterol biosensors prepared by layer-by-layer technique. *Biosens. Bioelectron.* 2001, *16*, 849–856.
- 162. Samanta, K.; Sarkar, A. Immobilization of bio-macromolecules on self-assembled monolayers: Methods and sensor applications. *Chem. Soc. Rev.* **2011**, *40*, 2567–2592.
- 163. Yang, M.; Kabulski, J.L.; Wollenberg, L.; Chen, X.; Subramanian, M.; Tracy, T.S.; Lederman, D.; Gannett, P.M.; Wu, N. Electrocatalytic drug metabolism by CYP2C9 bonded to a self-assembled monolayer-modified electrode. *Drug Metab. Dispos.* 2009, 37, 892–899.
- 164. De Micheli, G.; Ghoreishizadeh, S.S.; Boero, C.; Valgimigli, F.; Carrara, S. An Integrated Platform for Advanced Diagnostics, In *Proceedings of DATE 2011 Conference*, Grenoble, France, 14–18 March 2011; pp. 1–6.
- 165. Cassell, A.M.; Raymakers, J.A.; Kong, J.; Dai, H. Large scale CVD synthesis of single-walled carbon nanotubes. *J. Phys. Chem. B* **1999**, *103*, 6484–6492.
- Nessim, G.D.; Acquaviva, D.; Seita, M.; O'Brien, K.P.; Thompson, C.V. Adv. Funct. Mater. 2010, 20, 1306–1312.
- 167. Wang, H.; Feng, J.; Hu, X.; Ng, K. Synthesis of aligned carbon nanotubes on double-sided metallic substrate by chemical vapor deposition. *J. Phys. Chem. C* **2007**, *111*, 12617–12624.
- 168. Bayer, B.C.; Hofmann, S.; Castellarin-Cudia, C.; Blume, R.; Baehtz, C.; Esconjauregui, S.; Wirth, C.T.; Oliver, R.A.; Ducati, C.; Knop-Gericke, A.; *et al.* Support-catalyst-gas interactions during carbon nanotube growth on metallic Ta films. *J. Phys. Chem. C* 2011, *115*, 4359–4369.
- 169. Martin-Fernandez, I.; Gabriel, G.; Rius, G.; Villa, R.; Perez-Murano, F.; Lora-Tamayo, E.; Godignon, P. Vertically aligned multi-walled carbon nanotubes growth on platinum electrodes for bio-impedance applications. *Microelectron. Eng.* 2009, *86*, 806–808.
- 170. Talapatra, S.; Kar, S.; Pal, S.K.; Vajtai, R.; Ci, L.; Victor, P.; Shaijumon, M.M.; Kaur, S.; Nalamasu, O.; Ajayan, P.M. Direct growth of aligned carbon nanotubes on bulk metals. *Nat. Nano* 2006, *1*, 112–116.
- 171. Matthews, K.D.; Lemaitre, M.G.; Kim, T.; Chen, H.; Shim, M.; Zuo, J.M. Growth modes of carbon nanotubes on metal substrates. *J. Appl. Phys.* **2006**, *100*, doi:10.1063/1.2219000.
- 172. Lin, Y.; Lu, F.; Tu, Y.; Ren, Z. Glucose biosensors based on carbon nanotube nanoelectrode ensembles. *Nano Lett.* **2004**, *4*, 191–195.
- 173. Minati, L.; Speranza, G.; Torrengo, S.; Toniutti, L.; Migliaresi, C.; Maniglio, D. Ferrari, M.; Chiasera, A. Characterization of thiol-functionalized carbon nanotubes on gold surfaces. *Surf. Sci.* 2010, 604, 1414–1419.

- 174. Hu, C.; Hu, S. Carbon nanotube-based electrochemical sensors: Principles and applications in biomedical systems. *J. Sens.* **2009**, *2009*, 1–40.
- 175. Guiseppi-Elie, A.; Rub, A.; Rahman A.; Shukla N.K. SAM-modified microdisc electrode arrays (MDEAs) with functionalized carbon nanotubes. *Electrochim. Acta* **2010**, *55*, 4247–4255.
- 176. Zuaznabar-Gardona, J.C.; Cao, R.; Suarez, M. Vertical self-assembly of modified multiwalled carbon nanotubes on gold surfaces induced by chitosan and Tween. *Chem. Commun.* 2012, 48, 1910–1912.
- 177. Juan, X.J.; Gang, W.; Qing, X.H.; Zhang, X.I.A.; Hong, Y.C. Third generation horseradish peroxidase biosensor based on self-assembling carbon nanotubes to gold electrode surface. *Chin. Chem. Lett.* **2005**, *16*, 523–526.
- 178. Kim, H.R.; Park, S.; Jung, C.; Kim, J.; Rhee, C.K.; Hyun, M.S. Selective adsorption of dithiolate-modified multi-wall carbon nanotubes onto alkanethiol self-assembled monolayers on Au(111). *Chem. Commun.* **2010**, *46*, 6584–6586.
- 179. Yu, X.; Chattopadhyay, D.; Galeska, I.; Papadimitrakopoulos, F.; Rusling, J.F. Peroxidase activity of enzymes bound to the ends of single-wall carbon nanotube forest electrodes. *Electrochem. Commun.* **2003**, *5*, 408–411.
- 180. Lai, K.W.C.; Fung, C.K.M.; Li, W.J. A Systematic Approach to Fabricate CNT-Based Nano Devices: Combining DEP and Microspotting Technologies. In. *Proceedings of 5th IEEE Conference on Nanotechnology*, Nagoya, Japan, 15–18 July 2005; pp. 350–353.
- 181. Boero, C.; Carrara, S.; de Micheli, G. New Technologies for Nanobiosensing and Their Applications to Real-Time Monitoring, In *Proceedings of IEEE BioCAS 2011 Conference*, San Diego, CA, USA, 10–12 November 2011; pp. 357–360.
- 182. Lai, K.W.C.; Xi, N. Novel Carbon Nanotube Deposition System for Fabricating Nano Devices. In *Proceedings of 2nd IEEE Conference NEMS 2007*, Bangkok, Thailand, 16–19 January 2007; pp. 1081–1084.
- 183. Lai, K.W.C.; Fung, C.K.M.; Li, W.J. Fabrication of CNT Nanosensors by Combining Micro-Robotic Spotting and DEP Technologies. In *Proceedings of IEEE Workshop on Advanced Robotics and its Social Impacts*, Nagoya, Japan, 12–15 June 2005; pp. 30–35.
- 184. Ahmadi, M.M.; Jullien, G.A. A wireless-implantable microsystem for continuous blood glucose monitoring, biomedical circuits and systems. *Trans. IEEE* **2009**, *3*, 169–180.
- Haider M.R.; Islam, S.K.; Mostafa, S.; Mo, Z.; Taeho, O.H. Low-power low-voltage current readout circuit for inductively powered implant system. *IEEE Trans. Biomed. Circuits Syst.* 2010, *4*, 205–213.
- 186. Lin, L.; Qureshi, W.A.; Xiaowen L.; Mason, A.J. Amperometric Instrumentation System with On-Chip Electrode Array for Biosensor Application. In *Proceedings of the IEEE BioCAS 2010 Conference*, Paphos, Cyprus, 3–5 November 2010; pp. 294–297.
- 187. Roham, M. Daberkow, D.P.; Ramsson, E.S.; Covey, D.P.; Pakdeeronachit, S.; Garris, P.A.; Mohseni, P. A wireless IC for wide-range neurochemical monitoring using amperometry and Fast-scan cyclic voltammetry. *IEEE Trans. Biomed. Circuits Syst.* 2008, 2, 3–9.
- 188. Nambiar, N.; Blalock, B.J. A novel current-mode multi-channel integrating ADC. *Analog Integr.* Circuits. *Sign. Process.* **2010**, *63*, 283–291.

- 189. Zhu, X.; Ahn, C.H. On-chip electrochemical analysis system using nanoelectrodes and bioelectronic CMOS chip. *IEEE Sens. J.* **2006**, *6*, 1280–1286.
- 190. Ahmadi, M.M.; Jullien, G.A. Current-mirror based potentiostats for three-electrode amperometric electrochemical sensors. *IEEE Trans. Circuits Syst. I* 2009, *56*, 1339–1348.
- 191. Bard, A.J.; Faulkner, L.R. *Electrochemical Methods: Fundamentals and Applications*; Wiley: New York, NY, USA, 1980.
- 192. Razavi, B. *Design of Analog CMOS Integrated Circuits*, Chapter 12; McGraw-Hill: New York, NY, USA, 2001; pp. 410–423.
- 193. Rub, A.; Rahman, A.; Justin, G.; Guiseppi-Wilson, A.; Guiseppi-Elie, A. Fabrication and packaging of a dual sensing electrochemical biotransducer for glucose and lactate useful in intramuscular physiologic status monitoring. *IEEE Sens. J.* **2009**, *9*, 1856–1863.
- 194. De Venuto, D.; Torre, M.D.; Boero, C.; Carrara, S.; de Micheli, G. A Novel Multi-Working Electrode Potentiostat for Electrochemical Detection of Metabolites. In *Proceedings of the IEEE* Sensors 2010 Conference, Waikoloa, HI, USA, 1–4 November 2010; pp. 1572–1577.
- 195. Ghoreishizadeh, S.; Carrara, S.; de Micheli, G. Circuit Design for Human Metabolites Biochip. In *Proceedings of IEEE BioCAS 2011 Conference*, San Diego, CA, USA, 10–12 November 2011; pp. 460–463.
- 196. Briman, M.; Bradley, K.; Gruner, G. Source of 1/f noise in carbon nanotube devices. *J. Appl. Phys.* **2006**, *100*, doi:10.1063/1.2210570.
- 197. Yang, Z.; Zhao, Q.; Keefer, E.; Liu, W. Noise Characterization, Modeling, and Reduction for *In Vivo* Neural Recording. In *Proceeding of NIPS 2009 Conference*, Singapore, 11 December 2009; pp. 1–9.
- 198. Hassibi, A.; Navid, R.; Dutton, R.W.; Lee, T.H. Comprehensive study of noise processes in electrode electrolyte interfaces. J. Appl. Phys. 2004, 96, 1074–1082.
- 199. Mitcheson, P.D.; Yeatman, E.M.; Rao, G.K.; Holmes, A.S.; Green, T.C. Energy harvesting from human and machine motion for wireless electronic devices. *Proc. IEEE* **2008**, 96, 1457–1486.
- 200. Silay, K.M.; Dehollain, C.; Declercq, M. Improvement of Power Efficiency of Inductive Links for Implantable Devices. In *Proceedings of the PRIME 2008 Conference*, Istanbul, Turkey, 22–25 June 2008; pp. 229–232.
- 201. Silay, K.M.; Dehollain, C.; Declercq, M. Inductive power link for a wireless cortical implant with two-body packaging. *IEEE Sens. J.* **2011**, *11*, 2825–2833.
- 202. Finkenzeller, K. RFID Handbook, 3rd ed.; Wiley: New York, NY, USA, 2010.
- 203. Activa RC Recharging System 37751. Medtronic.
- 204. Olivo, J.; Carrara, S. de Micheli, G. IronIC Patch: A Wearable Device for the Remote Powering and Connectivity of Implantable Systems. In *Proceedings of IEEE 12MTC 2011 Conference*, Graz, Austria, 13–16 May 2012; pp. 1–4.
- 205. Kumosa, G.L.; Routh, T.; Lin, J.; Lucisano, J. Function of an implanted tissue glucose sensor for more than 1 year in animals. *Sci. Transl. Med.* **2010**, *2*, 42–53.
- 206. Frost, M.; Meyerhoff, M. Implantable chemical sensors for real time clinical monitoring: Progress and challenges. *Curr. Opin. Chem. Biol.* **2002**, *6*, 633–641.
- 207. Rather, B.D.; Hoffman, A.S.; Schoen, F.J.; Lemons, J.E. *Biomaterials Science: An Introduction to Materials in Medicine*, 2nd ed.; Elsevier Academic Press: New York, NY, USA, 2004.

- 208. Williams, D.F. *The Williams Dictionary of Biomaterials*; Liverpool University Press: Liverpool, UK, 1999.
- 209. Anderson, J.M.; Langone, J.J. Issues and perspectives on the biocompatibility and immunotoxicity evaluation of implanted controlled release systems. J. Control. Release 1999, 57, 107–113.
- 210. Biological Compatibility of Medical Devices. Part 5. Tests for Cytotoxicity: In Vitro Methods; ISO 10993; ISO: Geneva, Switzerland, 1992.
- Saha, K.; Pollock, J.F.; Schaffer, D.V.; Healy, K.E. Designing synthetic materials to control stem cell phenotype. *Curr. Opin. Chem. Biol.* 2007, 11, 381–387.
- 212. Sharkawy, A.A.; Klitzman, B.; Truskey, G.A.; Relchert, W.M. Engineering the tissue which encapsulates subcutaneous implants. II. Plasma-tissue exchange properties. J. Biomed. Mater. Res. 1998, 40, 586–597.
- 213. Popat, K.C.; Eltgroth, M.; Latempa, T.J.; Grimes, C.A.; Desai, T.A. Decreased *Staphylococcus epidermis* adhesion and increased osteoblast functionality on antibiotic-loaded titania nanotubes. *Biomaterials* **2007**, *28*, 4880–4888.
- Desai, T.A.; Leoni, L.; Essenpreis, M.; Ferrari, M. Nanoporous anti-fouling silicon membranes for biosensor applications. *Biosens. Bioelectron.* 2000, 15, 453–462.
- 215. Koschwanez, H.; Yap, F.Y.; Klitzman, B.; Reichert, W.M. *In Vitro* and *in vivo* characterization of porous poly-L-lactic acid coatings for subcutaneously implanted glucose sensors. *J. Biomed. Mater. Res. Part A* 2008, 87, 792–807.
- 216. Yu, B.; Wang, C.; Ju, Y.M.; West, L.; Harmon, J.; Moussy, Y.; Moussy, F. Use of hydrogel coating to improve the performance of implanted glucose sensors. *Biosens. Bioelectron.* 2008, 23,1278–1284.
- 217. Quinn, C.A.P.; Connor, R.E.; Heller, A. Biocompatible, glucose-permeable hydrogel for *in situ* coating of implantable biosensors. *Biomaterials* **1997**, *18*, 1665–1670.
- 218. Quinn, C.P. Photo-crosslinked copolymers of 2-hydroxyethyl methacrylate, poly (ethylene glycol) tetra-acrylate and ethylene dimethacrylate for improving biocompatibility of biosensors. *Biomaterials* **1995**, *16*, 389–396.
- 219. Ward, W.K. The effect of local subcutaneous delivery of vascular endothelial growth factor on the function of a chronically implanted amperometric glucose sensor. *Diabetes Technol. Ther.* 2004, *6*, 137–145.
- 220. Galeska, I. Controlled release of dexamethasone from PLGA microspheres embedded within polyacid-containing PVA hydrogels. *AAPS J.* **2005**, *7*, 231–240.
- 221. Hickey, T., Kreutzer, D.; Burgess, D.J.; Moussy, F. *In vivo* evaluation of a dexamethasone/PLGA microsphere system designed to suppress the inflammatory tissue response to implantable medical devices. *J. Biomed. Mater. Res.* **2002**, *61*, 180–187.
- 222. Op de Beeck, M.; La Manna, A.; Buisson, T.; Dy, E.; Velenis, D.; Axisa, F.; Soussan, P.; van Hoof, C. An IC-Centric Biocompatible Chip Encapsulation Fabrication Process. In *Proceedings of 2010 3rd Electronic System-Integration Technology Conference (ESTC)*, Berlin, Germany, 13–16 September 2010.
- 223. Op de Beeck, M.; Qian, K.; Fiorini, P.; Malachowski, K.; van Hoof, C. Design and Characterization of A Biocompatible Packaging Concept for Implantable Electronic Devices. In *Proceedings of 44th International Symposium on Microelectronics*, Long Beach, CA, USA, 9–13 October 2011.

- 224. Dy, E. Biocompatibility Assessment of Advanced Wafer-Level Based Chip Encapsulation 2010. In *Proceedings of IEEE ESTC 2010 Conference*, Berlin, Germany, 13–16 September 2010; pp. 1–4.
- 225. Rodger, D. High-Density Flexible Parylene-Based Multielectrode Arrays for Retinal and Spinal Cord Stimulation 2007. In *Proceedings of Transducers 2007 Conference*, Lyon, France, 10–14 June 2007; pp. 1385–1388.
- 226. Yao, Y.; Gulari, M.N.; Casey, B.; Wiler, J.A.; Wise, K.D. Silicon Microelectrodes with Flexible Integrated Cables for Neural Implant Applications 2007. In *Proceedings of IEEE/EMBS 2007 Conference*, Cambridge, UK, 19–22 August 2007; pp. 398–401.
- 227. Stieglitz, T.; Schuetter, M.; Koch, K.P. Implantable biomedical microsystems for neural prostheses. *IEEE Eng. Med. Biol. Mag.* 2005, *24*, 58–65.
- 228. Hetke, J.F.; Lund, J.L.; Najafi, K.; Wise, K.D.; Anderson, D.J. Silicon ribbon cables for chronically implantable microelectrode arrays. *Biomed. Eng. IEEE Trans.* **1994**, *41*, 314–321.
- 229. Kotzar, G.; Freas, M.; Abel, P.; Fleischman, A.; Roy, S.; Zorman, C.; Moran, J.M.; Melzak, J. Evaluation of MEMS materials of construction for implantable medical devices. *Biomaterials* 2002, 23, 2737–2750.
- 230. Topper, M.; Klein, M.; Buschick, K.; Glaw, V.; Orth, K.; Ehrmann, O.; Hutter, M.; Oppermann, H.; Becker, K.-F.; Braun, T.; *et al.* Biocompatible Hybrid Flip Chip Microsystem Integration for Next Generation Wireless Neural Interfaces. In *Proceedings of 56th Electronic Components and Technology Conference*, San Diego, CA, USA, 5 July 2006; p. 4.
- 231. Kyrolainen, M.; Rigsby, P.; Eddy, S.; Vadgama, P. Bio-/haemocompatibility: Implications and outcomes for sensors? *Acta Anaesthesiol. Scand.* **1995**, *39*, 55–60.
- 232. Moussy, F.; Harrison, D.J. Prevention of the rapid degradation of subcutaneously implanted Ag/AgCl reference electrodes using polymer coatings. *Anal. Chem.* **1994**, *66*, 674–679.
- 233. Labat-Allietta, N.; Thévenot, D.R. Influence of calcium on glucose biosensor response and on hydrogen peroxide detection. *Biosens. Bioelectron.* **1998**, *13*, 19–29.
- 234. Silber, A., Hampp, N.; Schuhmann, W. Poly (methylene blue)-modified thick-film gold electrodes for the electrocatalytic oxidation of NADH and their application in glucose biosensors. *Biosens. Bioelectron.* **1996**, *11*, 215–223.
- 235. Wisniewski, N.; Reichert, M. Methods for reducing biosensor membrane biofouling. *Colloids Surf. B Biointerfaces* **2000**, *18*, 197–219.
- 236. Akyildiz, F.; Su, W.; Sankarasubramaniam, Y.; Cayirci, E. Wireless sensor networks: A survey. *Comput. Netw.* **2002**, *38*, 393–422.
- 237. Anderson, R. Security Engineering: A Guide to Building Dependable Distributed Systems, 2nd ed.; Wiley: New York, NY, USA, 2008.
- 238. Paul, N.; Kohno, T.; Klonoff, D. A review of the security of insulin pump infusion systems. *J. Diabetes Sci. Technol.* **2011**, *5*, 1557–1562.
- 239. Halperin, D.; Heydt-Benjamin, T.; Ransford, B.; Clark, S.; Defend, B.; Morgan, W.; Fu, K.; Kohno, T.; Maisel, W. Pacemakers and Implantable Cardiac Defibrillators: Software Radio Attacks and Zero-Power Defenses. In *Proceedings of IEEE Symposium on Security and Privacy*, Berkeley/Oakland, CA, USA, 5–8 May 2008; pp. 579–582.
- 240. Guiseppi-Elie, A. An implantable biochip to influence patient outcomes following trauma-induced hemorrhage. *J. Anal. Bioanal. Chem.* **2011**, *399*, 403–419.

- 241. Rabaey, J.M.; Mark, M.; Chen, D.; Sutardja, C.; Tang, C.; Gowda, S.; Wagner, M.; Werthimer, D. Powering and Communicating with mm-Size Implants. In *Proceedings of IEEE DATE 2011 Conference*, Grenoble, France, 14–18 March 2011; pp. 1–6.
- 242. Masny, B.D.; Geiselmann, W.; Bretthauer, G. Block Cipher Based Security for Severely Resource-Constrained Implantable Medical Devices. In *Proceedings of 4th ISABEL Conference* 2011, Barcellona, Spain, 26–29 October 2011; pp. 111–115.
- 243. Huang R.; Suh, E. IVEC: Off-Chip Memory Integrity Protection for Both Security and Reliability. In Proceedings of ISCA Conference 2010, Saint-Malo, France, 19–23 June 2010; pp. 395–406.
- 244. Win, K.; Susilo, W. Personal health record systems and their security protection. *J. Med. Syst.* **2006**, *30*, 309–315.
- 245. Hosseini-Khayat, S. A Lightweight Security Protocol for Ultra-Low Power ASIC Implementation for Wireless Implantable Medical Devices. In *Proceedings of ISMICT Conference 2011*, Montreux, Switzerland, 27–30 March 2011; pp. 554–557.
- 246. Guan, S.; Gu, J.; Shen, Z.; Wang, J.; Huang, Y.; Mason, A. A Wireless Powered Implantable Bio-Sensor Tag System-on-Chip for Continuous Glucose Monitoring. In *Proceedings of IEEE BioCAS 2011 Conference*, San Diego, CA, USA, 10–12 November 2011; pp. 193–196.
- 247. Gollakota, S.; Hassanieh, H.; Ransford, B.; Katabi, D.; Fu, K. They Can Hear Your Heartbeats: Non-Invasive Security for Implanted Medical Devices. In *Proceedings of ACM SIGCOMM 2011 Conference*, Toronto, ON, Canada, 15–19 August 2011; pp. 345–450.
- 248. Sorber, J.; Shin, M.; Peterson, R.; Cornelius, C.; Mare, S.; Prasad, A.; Marois, Z.; Smithayer, E.; Kotz, D. An Amulet for Trustworthy Wearable mHealth. In *Workshop on Mobile Computing Systems and Applications* (HotMobile); San Diego, CA, USA, 2–3 February 2012; pp. 1–6.
- 249. Cornelius, C.; Kotz, D. Recognizing Whether Sensors are on the Same Body. In *Proceedings of the 9th International Conference on Pervasive Computing*, San Francisco, CA, USA, 12–15 June 2011; pp. 332–349.
- 250. Burleson, W. Physical Layer Security and Privacy with Ultra-Wideband. In *Proceedings of ISMICT Conference 2011*, Montreux, Switzerland, 27–30 March 2011; pp. 14–18.
- 251. Danev, B.; Heydt-Benjamin, T.; Capkun, S. Physical-Layer Identification of RFID Devices. In *Proceedings of USENIX Security Symposium 2009*, Montreal, QC, Canada, 10–14 August 2009; pp. 1–16.
- 252. Burleson, W.; Clark, S.; Ransford, B.; Fu, K. Design Challenges in Secure Implantable Medical Devices. In *Proceedings of ACM/IEEE DAC 2012 Conference*, San Francisco, CA, USA, 3–7 June 2012; pp. 1–6.
- 253. Baj-Rossi, C.; de Micheli, G.; Carrara, S. Electrochemical detection of anti-breast-cancer agents in human serum by cytochrome P450-coated carbon nanotubes. *Sensors* **2012**, *12*, 6520–6537.
- 254. Carrara, S. Bio/CMOS Interfaces and Co-Design; Springer: New York, NY, USA, 2012.

© 2012 by the authors; licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution license (http://creativecommons.org/licenses/by/3.0/).